@AGUPUBLICATIONS

Water Resources Research



10.1002/2017WR020806

Key Points:

- SY should be modeled in ESMs at the event scale with including erosion processes more than interrill and rill erosion
- Seismicity and land management that are rarely considered in current models could be important for controlling global SY
- A statistically significant empirical relationship between sediment and POC yield has been established for simulating them jointly in ESMs

Supporting Information:

Supporting Information S1

Data Set S1

Correspondence to:

L. R. Leung, Ruby.Leung@pnnl.gov

Citation:

Tan, Z., Leung, L. R., Li, H., Tesfa, T., Vanmaercke, M., Poesen, J., . . . Hartmann, J. (2017). A Global data analysis for representing sediment and particulate organic carbon yield in Earth System Models. *Water Resources Research*, *53*, 10,674–10,700. https:// doi.org/10.1002/2017WR020806

Received 21 MAR 2017 Accepted 27 NOV 2017 Accepted article online 1 DEC 2017 Published online 21 DEC 2017

A Global Data Analysis for Representing Sediment and Particulate Organic Carbon Yield in Earth System Models

Zeli Tan¹ (b), L. Ruby Leung¹ (b), Hongyi Li² (b), Teklu Tesfa¹, Matthias Vanmaercke³, Jean Poesen⁴, Xuesong Zhang¹, Hui Lu⁵ (b), and Jens Hartmann⁶ (b)

¹Pacific Northwest National Laboratory, Richland, WA, USA, ²Montana State University, Bozeman, MT, USA, ³Département de Géographie, Université de Liège, Liege, Belgium, ⁴Department of Earth and Environmental Sciences, Division of Geography, KU Leuven, Leuven, Belgium, ⁵Ministry of Education Key Laboratory for Earth System Modeling, Department of Earth System Science, Tsinghua University, Beijing, China, ⁶Institute for Geology, Center for Earth System Research and Sustainability, Universität Hamburg, Hamburg, Germany

Abstract Although sediment yield (SY) from water erosion is ubiquitous and its environmental consequences are well recognized, its impacts on the global carbon cycle remain largely uncertain. This knowledge gap is partly due to the lack of soil erosion modeling in Earth System Models (ESMs), which are important tools used to understand the global carbon cycle and explore its changes. This study analyzed sediment and particulate organic carbon yield (CY) data from 1,081 and 38 small catchments (0.1–200 km²), respectively, in different environments across the globe. Using multiple statistical analysis techniques, we explored environmental factors and hydrological processes important for SY and CY modeling in ESMs. Our results show clear correlations of high SY with traditional agriculture, seismicity and heavy storms, as well as strong correlations between SY and annual peak runoff. These highlight the potential limitation of SY models that represent only interrill and rill erosion because shallow overland flow and rill flow have limited transport capacity due to their hydraulic geometry to produce high SY. Further, our results suggest that SY modeling in ESMs should be implemented at the event scale to produce the catastrophic mass transport during episodic events. Several environmental factors such as seismicity and land management that are often not considered in current catchment-scale SY models can be important in controlling global SY. Our analyses show that SY is likely the primary control on CY in small catchments and a statistically significant empirical relationship is established to calculate SY and CY jointly in ESMs.

Plain Language Summary Sediment and organic carbon in the rivers produced by soil erosion are ubiquitous. Although they have important effects on the global carbon cycle, current models have limitations in representing sediment and particulate organic carbon (POC) yield at temporal and spatial scales relevant to Earth System Models (ESMs). By analyzing the sediment yield data from over 1000 small catchments across the globe, we identified environmental factors and hydrological processes important for modeling sediment yield in ESMs. Based on the POC yield data, we indicated that sediment yield is likely the primary control on POC yield. Importantly, we also established a statistical significant empirical relationship relating POC yield to sediment yield that can be used in ESMs.

1. Introduction

Recent studies have refuted the traditional view that aquatic ecosystems play a passive role in the global carbon cycle as a "pipe" to transport organic carbon from land to oceans. Increasingly, observational evidence suggests that a considerable fraction of carbon in the aquatic ecosystems is biogeochemically modified in its passage to the oceans (Aufdenkampe et al., 2011; Cole et al., 2007; Kempe, 1982, 1984; Raymond et al., 2013; Richey et al., 2009). Therefore, estimating the magnitude and timing of carbon fluxes from land to rivers is critical for understanding how the carbon cycle responds to perturbations (Regnier et al., 2013) and for modeling large-scale riverine biogeochemistry in Earth System Models (ESMs; Lal, 2003). More specifically, particulate organic carbon (POC) fluxes from land to rivers through sediment yield (SY) should be accounted for.

© 2017. American Geophysical Union. All Rights Reserved. SY is a process that involves soil erosion, sediment transport, and deposition and is defined by the amount of sediment per unit basin area reaching a river basin outlet during a given period (units: t km⁻² yr⁻¹; Dialynas et al., 2016; Hilton et al., 2015; Lal, 2003; Martin et al., 2013). SY is important in riverine biogeochemistry because the amount of POC detached in erosion and sediment transport can be large. Assuming 45% of the eroded organic carbon is in particulate form (Ludwig et al., 1996), Lal (2003) estimated that the yield of POC could be as large as 0.9-1.35 pg C yr⁻¹, which is as much as 13.5% of the global net ecosystem production (10 pg C yr⁻¹) by terrestrial ecosystems (Watson et al., 2000). Furthermore, the detached sediment and POC can significantly affect the function of aquatic ecosystems in rivers by changing the river morphology, turbidity, and biogeochemistry (Lacoul & Freedman, 2006; Valero-Garcés et al., 1999; Verstraeten & Poesen, 2002; Vihermaa et al., 2014). Lastly, the erosion and sediment transport processes that produce sediment and POC fluxes are highly sensitive to climate change and human disturbance (Bork et al., 2001; Li & Fang, 2016; Nachtergaele et al., 2001; Syvitski et al., 2005; Vanmaercke et al., 2015, 2016), so gaining a physical understanding of these process is critical for predicting their changes in the future.

Because the dynamics of SY is dependent on catchment area (scale) and catchment characteristics that vary spatially and temporally (heterogeneity), modeling SY at the global scale is challenging. Hence, SY is rarely represented in ESMs although different modeling approaches have been developed for simulating SY in specific catchments (Yu et al., 1999) and global large river basins (Syvitski & Milliman, 2007). According to Poesen et al. (1996), Renschler and Harbor (2002), and others, soil detachment is scale dependent as it could be dominated by splash and interrill erosion at the microscale (mm to m) and plot scale (m to 100 m), rill erosion at the field scale (100–10,000 m), and channel processes (e.g., gully erosion) and mass movement (e.g., landsliding) at the hillslope-catchment scale (>10,000 m). Sediment deposition could be more important in large catchments and river basins due to limited transport capacity (Foster, 1982; Meade et al., 1985). The scale issue is also critical as it influences data availability and quality for modeling (Renschler & Harbor, 2002): high-quality data are less available in small-size fields, entire hillslopes, and small catchments.

Heterogeneity represents another aspect that challenges SY modeling. For different catchments, different environmental factors may dominate the SY. They include topography, land cover, soil texture, soil moisture, soil organic carbon (SOC), rock fragment, lithology, and tectonics (Bryan, 2000; Dialynas et al., 2016; Knapen et al., 2007; Poesen et al., 1994; Renschler & Harbor, 2002; Syvitski & Milliman, 2007; Vanmaercke et al., 2015). Additionally, because land management strongly influence SY (Syvitski et al., 2005; Vanmaercke et al., 2015), there can be large differences between natural catchments and catchments affected by humans.

In the past decades, many studies have published observations of SY across the globe. Synergistic analysis of published data may yield more generalizable findings on SY that are much needed to advance global modeling. Recently, several studies have emerged in this direction to identify significant hydrological processes or environmental factors consistently at the continental or global scale (Mutema et al., 2015; Vanmaercke et al., 2014a, 2014b, 2015). However, some aspects of these studies may limit their values for informing SY modeling in ESMs. For instance, these studies usually did not focus on spatial scales that are relevant to ESMs, and only data with sparse spatial coverage were included in the analysis at the global scale (Mutema et al., 2015). In addition, although observations indicated that event-scale rainfall and runoff are more important than annual rainfall and runoff in determining SY (Beven et al., 2005; Bryan, 2000; Warrick et al., 2015), few previous studies have included these event-scale variables in their analyses. Further, none of these studies focused on linking POC fluxes with sediment fluxes in catchments resolved by ESMs, a relationship that has been established for large river basins (Galy et al., 2015).

This study aims to identify environmental factors that should be represented in ESMs in order to simulate SY globally. It also aims to explore the direction of implementing SY modeling in ESMs. To fill the aforementioned gaps, we analyzed sediment and POC yield data collected from 1,081 and 38 small catchments with a size range of 0.1–200 km² across the globe, respectively. This size range is comparable to the spatial scale of processes represented in the land and river components of ESMs that have a typical grid resolution of 15–150 km with subgrid parameterizations of fine-scale processes, such as hillslope water dynamics (Li et al., 2013). Although there are gaps in the geographical coverage, the large number of catchments represent different climate zones, topography, mineralogy, geology, lithology, and land covers, which govern important aspects of the spatial variability of SY globally. As our focus is on upstream catchments, data that

were analyzed include low-order streams in the Mediterranean countries, East Africa, New Zealand, the Andes, Puerto Rico and the Southwest of China where SY is very high (Pelletier et al., 2015).

Overall, this study addresses four specific science questions relevant to modeling the yield of sediment and POC in ESMs: (1) What common natural and anthropogenic environmental factors are important for determining SY across different environments globally? (2) Is representing only interrill and rill erosion, as typical in existing SY models, sufficient for SY modeling in ESMs? (3) What is the appropriate time scale to simulate SY processes in ESMs? (4) What control POC fluxes in small catchments resolvable by ESMs? Section 2 describes the data and analysis methods. Results from the synthetic analysis are discussed in section 3. Section 4 discusses the implications of our results and limitations of the study, and section 5 summarizes the key findings and the necessary future studies.

2. Materials and Methods

2.1. Data Sources

Sediment and POC yield data (Figure 1) were obtained from various sources including values published in the literature and data sets (Table 1 and Appendix A). For instance, SY data for the United States were mainly derived from the USGS National Water Information System (NWIS; http://waterdata.usgs.gov/nwis). SY data for Canada, China, and Africa were obtained from the HYDAT database (http://collaboration.cmc.ec. gc.ca/cmc/hydrometrics/www/), the Chinese Hydrology Data Project (Henck et al., 2010, 2011), and the SY in Africa data set (Vanmaercke et al., 2014b), respectively. SY data for Europe were mainly compiled by Vanmaercke et al. (2011) for catchments less than 200 km² in size, with reliable outlet coordinates. Other SY data and most of the POC yield (CY) data were obtained from published articles through Google Scholar searches for relevant articles published in peer-reviewed journals in the field. Our search terms include "soil erosion," "sediment yield," "sediment loading," "sediment fluxes," "organic carbon loss," and "organic carbon fluxes." We only included values reported by articles that provided the outlet coordinate and observation period (no shorter than 1 year). The data sets and their detailed references can be found in the supporting information.

For SY that was reported more than 1 year, the data of each year were treated as a separate and independent measurement for temporal analysis. For spatial analysis, mean annual SY of each catchment was used. It was calculated by averaging the observed annual SY over the observational years. The averaging forces an equal weighting for each catchment to guarantee data independence in spatial analysis. Our method is different from Mutema et al. (2015), who treated data from multiple years for the same field as independent data. Figure 1 shows the locations of the catchments with SY and CY data provided at different temporal frequencies and supporting information Figures S1 and S2 show the observed mean annual SY and CY of



Figure 1. Outlet locations of the catchments with sediment yield (SY) or POC yield (CY) observations used in the study. Symbols indicate the duration and frequency of the SY and CY observations.

Table 1

Description of Response and Environmental Variables

Variable	Group	Description	Derived from	Range	Units
SY		Catchment sediment yield	Publications, data sets or personal communication	0.001-30,000	t km ^{-2} yr ^{-1}
CY		Catchment POC yield	Publications	0.0001-222.9	t km $^{-2}$ yr $^{-1}$
CA	Scale	Catchment area	Same as SY	0.01-200	km ²
Q	Climate	Mean annual runoff	Publications, USGS NIWS, Canada HYDAT, or a 50 year 0.25° global data set of surface hydrology simulated by Variable Infiltration Capacity (VIC) model (Sheffield & Wood, 2007)	0.6–7,208	mm
Q_{peak}	Climate	Annual peak runoff	Publications, USGS NIWS, or Canada HYDAT	0.0001–1,801	mm
Р	Climate	Mean annual precipitation	Publications or a 50 year 3-hourly 0.25° global data set of meteorological forcings for land surface modeling (Sheffield et al., 2006)	75–9,500	mm
RE	Climate	Average rainfall erosivity	A 50 year 3-hourly 0.25° global data set of meteorological forcings for land surface modeling (Sheffield et al., 2006)	1,135–55057	MJ mm ha ^{-1} h ^{-1} yr ^{-1}
VarSM	Moisture	Soil moisture variability that is defined as the relative difference between the highest and lowest soil moisture, divided by the average soil moisture	A 0.25° global monthly soil moisture data set retrieved from AMSR-E, covering the period from 2002 June to 2011 July (Lu et al., 2009)	1.7–249.1	%
SLP	Topography	Average catchment slope	Publications or a 90 m resolution DEM (Lehner et al., 2008)	0.002-0.602	$\mathrm{km}\mathrm{km}^{-1}$
RSLP	Topography	Average river channel slope	Same as SLP	0.0001-0.306	km km ⁻¹
PGA	Topography	Peak ground acceleration (a short-period ground motion parameter for the measure of ground movement amplitude) with an exceedance probability of 10% in 50 years	A 0.1° resolution global seismic hazard map (Giardini et al., 1999)	0–9.78	m s ⁻²
Frag	Topography	Topsoil coarse fragments volumetric content	A 1 km resolution global soil data product (Hengl et al., 2014)	0–57	%
Wood	Land use	Woodland coverage	Publications or a 0.05° global gridded land cover data set (Ke et al., 2012)	0–100	%
Grass	Land use	Grassland coverage	Same as wood	0-100	%
Crop	Land use	Cropland coverage	Same as wood	0-100	%
Clay	Soil	Topsoil clay content	Publications or a 1 km resolution global soil data product (Hengl et al., 2014)	3–63	%
Silt	Soil	Topsoil silt content	Same as clay	1–72	%
Sand	Soil	Topsoil sand content	Same as clay	0–90	%
SOC	Soil	Topsoil SOC content	Same as clay	0.5-41.7	%
LITH	Lithology	Lithology erodibility factor, defined by Syvitski and Milliman (2007) that was assigned to each lithology of a global lithology map depending on their erodibility	A 0.1° global gridded lithology map (Dürr et al., 2005)	0.1–3.0	N.A.
POP	Population	Population density of the catchment	A 2.5 arc min gridded population of the world data (GPW) version 3 (Center for International Earth Science Information Network Columbia University & Centro Internacional de Agricultura Tropical, 2005)	0–2,925	people km ⁻²

Note. "N.A." indicates not applicable. Variables are grouped based on the Spearman correlations in supporting information Table S1.

the study catchments, respectively. The U.S., Europe, and East Africa are the regions with higher spatial coverage. The SY data included in this study were measured either at a gauging station (N = 800) or calculated from reservoir siltation rates (N = 281) over a measuring period of at least 1 year. It should be noted that the SY estimated from reservoir siltation rates could be systematically higher than the estimates from gauging stations (Vanmaercke et al., 2011, 2014b) for two reasons: (1) reservoir surveys usually account for SY from both suspended load and bed load and (2) reservoir surveys overall better represent the importance of infrequent erosion events that often have a large impact on average SY, but that are often missed in gauging station measurements.

2.2. Definition of Variables

The units of sediment and POC yield (expressed as SY and CY, respectively) used here is t $\text{km}^{-2} \text{ yr}^{-1}$. For data reported as discharge-weighted sediment concentration values, they were converted to yield per unit area using the known outlet runoff discharge and catchment area (CA) values.

Earlier experiments and field investigations identified various environmental factors that could affect sediment and POC yield (Bryan, 2000; Dialynas et al., 2016; Knapen et al., 2007; Ludwig et al., 1996; Poesen et al., 1994; Renschler & Harbor, 2002; Syvitski & Milliman, 2007; Vanmaercke et al., 2015). In this study, we included annual precipitation (P), annual runoff (Q), mean catchment slope (SLP), mean channel slope (RSLP), woodland fraction (Wood), grassland fraction (Grass), cropland fraction (Crop), topsoil (top 10 cm), clay content (Clay), topsoil silt content (Silt), topsoil sand content (Sand), topsoil rock fragment content (Frag), topsoil organic carbon content (SOC), relative monthly variability of soil moisture (VarSM), peak ground acceleration with an exceedance probability of 10% in 50 years (PGA), lithology erodibility factor (LITH), population (POP), and CA in our analysis. To investigate the effects of event-scale rainfall and runoff, we included two event-based environmental variables (Griffiths, 1981; Gonzalez-Hidalgo et al., 2010; López-Tarazóna et al., 2010; Renard et al., 1997; Tropeano, 1991): annual peak runoff (Qpeak) and rainfall erosivity (RE). RE is defined as the sum of the kinetic energy of raindrops, which is a function of rainfall intensity (Dunne et al., 2016; Poesen & Savat, 1981). The definition, data source, data range, and units of these variables are listed in Table 1. Based on the spatial correlations in supporting information Table S1, we classified the environmental variables into eight groups: "Scale," "Climate," "Topography," "Land use," "Soil," "Lithology," "Moisture," and "Population" (Table 1).

In most catchments, the data sources did not report all of the required environmental information. For them, we calculated the average values of the environmental factors within their boundaries using several widely used data sets. For each catchment, watershed boundaries were delineated using the Terrain Analysis Using Digital Elevation Model (TauDEM) tools (Tesfa et al., 2011; Wallis et al., 2009) from a global watershed data set that was developed based on the 90 m Digital Elevation Model (DEM) from the Hydrological data and maps based on Shuttle Elevation Derivatives at multiple Scale (HydroSHEDS) database (Lehner et al., 2008). In this study, a noncropland catchment is defined as a catchment where the areal fraction of arable land is less than 20% (Vanmaercke et al., 2015). The soil management status of the U.S. cropland catchments was mainly identified using a county-level data set of tillage practices compiled by the Conservation Technology Information Center (CTIC, 2008). This data set was reprocessed into three categories: conventional tillage, conservation tillage, and no-till and gap-filled for 2000–2008 (West et al., 2010). For cropland catchments outside of the U.S., they are recorded as "managed" only if there are specific land management activities (e.g., reduce tillage) reported on the catchments in publications. As shown in Figures 2 and 3, the data sets used in our study cover a broad range of SY and catchment characteristics at the global scale, so they can be used in comprehensive analyses to determine the environmental factors that influence SY and CY consistently across the globe. Figures 2 and 3 also show clearly the skewness of the variables. Like the SY data, only the average values of multiple-year data for each catchment were used in spatial analysis of CY.

2.3. Statistical Analysis

To address our first three questions, the common environmental factors important for determining SY across different catchments and the role of rill and interrill erosion and event-scale erosion were analyzed. To identify the spatial correlations between environmental factors and between environmental factors and SY, we used linear regression and simple principal component analysis (PCA). Linear regression and scatterplots were used to disclose the simple linear correlation and complex nonlinear correlations between the control factors and SY, respectively (Bryan, 2000; Hassan et al., 2008; Vanmaercke et al., 2014b, 2015). The scatterplots were also used to show the degree of consistency between the catchments with Q_{peak} available and all other catchments. For severely skewed variables (e.g., SLP and PGA) shown in Figures 2 and 3, they were log-transformed before being analyzed using linear regression and PCA (Yang et al., 2016) because these analyses only apply to variables with the normal distribution. In the PCA, all variables were *z*-



Figure 2. Relative frequency distribution of catchment characteristics, including variables related to catchment scale (CA), topography (SLP, RSLP, PGA, and Frag), and soil (Clay, Silt, Sand, and SOC). The number of catchments involved in each analysis is also presented.

standardized and principal components (PCs) were calculated. Each PC is linearly uncorrelated in the spatial domain and can explain a certain percentage of the spatial variance of environmental factors, called component scores. The first two PCs that have the largest component scores, representing the dominant variance of environmental variables, were used in further analysis. Additionally, by projecting the SY vector onto the PCs, we can compare the dependence of SY on the two PCs (Mutema et al., 2015). The validity of the PCA was tested using the Bartlett tests of sphericity (p < 0.001) and the Kaiser-Meyer-Olkin measures of sampling adequacy (Snedecor & Cochran, 1989).

To investigate the impacts of human activities on SY, we used the one-way analysis of variance (ANOVA). The one-way ANOVA test is widely used to check the existence of significant differences between more than two data samples. In ANOVA, we tested the two-tier hypotheses that (1) the mean SYs of the three catchment groups with different land management activities are equal and (2) if the first hypothesis is rejected based on the *F*-statistics (p < 0.05), the mean SYs of any two catchment groups are equal based on the Tukey's honest significant difference test.

Figure 3. Relative frequency distribution of catchment characteristics, including variables related to land use (Wood, Grass, and Crop), population (POP), lithology (LITH), moisture (VarSM), and climate (P and Q), and relative frequency distribution of sediment yield (SY). The number of catchments involved in each analysis is also presented.

The complex correlations between environmental variables make it difficult to evaluate the important factors in controlling SY across the globe. The partial *r* coefficient and the corresponding *p*-value, calculated by controlling the interfactor effects from less correlated variables (Vanmaercke et al., 2014b), might be instructive on the relative significance of the association of SY with environmental variables, especially the relative significance of highly correlated variables. For the partial *r*, the highly correlated variables were identified and grouped based on the full Spearman correlation matrix of environmental variables (supporting information Table S1), with no requirement for transformation of the skewed variables. Since we are particularly interested in examining the explanatory power of event-scale variables (e.g., Q_{peak} and RE) on the spatial and temporal variability of SY relative to that of Q and P, the partial *r* was calculated from catchments with Q_{peak} and RE available. For the temporal variability of SY, the partial *r* was calculated from the catchments that have at least three measurements of SY in different years. Further, it should be expected that the partial *r* for temporal correlations would not be affected by many variables that do not vary temporally, such as soil texture.

Lastly, to address the fourth question of what control POC fluxes, multiple linear regression (MLR) models were used to establish the empirical relationships for SY and CY. This analysis can also help evaluate the importance and independence of the environmental factors in controlling SY at the global scale. To construct the optimal MLR of SY, we first employed an ensemble (10,000 times) of backward stepwise model selection (Vanmaercke et al., 2014b) to select environmental variables that have significant *F*-statistics (p < 0.05) for different subsets of the study catchments with Q_{peak} available (each subset includes half of the study catchments). Then, we made a single backward stepwise selection using the whole data set to further filter out variables that have less independent explanatory power on the spatial variability of SY. This way, the constructed MLR is less likely to be biased by influential data. For the optimal MLR of CY, because there are limited CY measurements, we only employed a single backward stepwise selection involving all environmental variables and SY. The variables with significant *F*-statistics were selected. As described in section 2.2, all of the above tests except the partial *r* for temporal correlations were based on the average values of the variables in Table 1 for each catchment if the catchment has multiple measurements in different years.

3. Results and Interpretations

3.1. Effects of Environmental Factors on SY

At the global scale, when only the main effect is considered using simple linear regressions (Figures 4 and 5), factors including Clay, Sand, PGA, POP, and VarSM show the strongest spatial correlations with the natural logarithm of the observed SY. More specifically, In(SY) is positively correlated with Clay, PGA, and POP and negatively correlated with Sand and VarSM. The spatial variability of all environmental factors can be best represented by two PCs in the PCA (Figure 6), with the first PC mainly corresponding to the variability of topography and the second PC mainly corresponding to the variability of climate. Not surprisingly, none of the factors can explain the spatial variability of SY at the global scale as predominantly as those reported in regional-scale analyses or large river basins (Syvitski & Milliman, 2007; Vanmaercke et al., 2014a, 2014b, 2015). But among the study factors, the relationships of PGA, POP, and Wood with SY are almost consistent with the previously reported relationships. The influence of seismic activity (expressed as PGA) on SY (Figure 4 and Table 2) is well documented in different regions (Broeckx et al., 2016; Dadson et al., 2003; Hovius, 1998; Vanmaercke et al., 2014a, 2014b). The weaker correlation between PGA and SY found in this study could be caused by a large fraction of the study catchments being located in regions with low seismic activities, e.g., the U.S. Midwest (Giardini et al., 1999). As shown in Figure 4 and Table 2, PGA has a closer correlation with SY than SLP in the analyses of linear regression and partial r. As documented by Hovius (1998), in very active mountain belts, slope/relief can reach a threshold point that an increase in uplift rate beyond the threshold causes further increases in SY without a corresponding increase in slope. It is also possible that active seismicity can weaken substrates/lithologies (Molnar et al., 2007) or cause landsliding (Dadson et al., 2003). Thus, there are very high erosion rates even in regions where topographic relief is low. It is not surprising that POP is positively correlated to SY because human activities (e.g., agriculture, humaninduced wildfire, infrastructure construction, and mining) are known to increase SY significantly. The effects of human activities on SY will be discussed in section 3.2. In contrast, because the presence of trees is known to protect soils from both rainfall-driven and runoff-driven erosion (Renard et al., 1997), Wood is negatively correlated to SY globally.

The positive correlation between Clay and SY as shown in Figure 4 is particularly interesting, given that many soil erosion models use clay content as a control factor on erosion even though there is no consensus on its effect. In some models, high clay content in soils is simulated to reduce rill erosion (Flanagan & Nearing, 1995; Renard et al., 1997), as clay content can increase soil aggregation and cohesion (Knapen et al., 2007; Panagos et al., 2014). In other models that include the Clay effect, high soil clay content is simulated to increase SY (Cohen et al., 2009; Coulthard et al., 1996; Temme & Vanwalleghem, 2016), as the occurrence of rill erosion, piping erosion and landsliding was observed to be high in clay soils (Broeckx et al., 2016; Gilley et al., 1993; Verachtert et al., 2011). Here we noted from Figure 6 that the projection of clay fraction and slope on the first PC lies on the opposite side of each other but the projection of clay fraction and runoff on the second PC lie on the same side. So it is possible that the positive correlation between Clay and SY is mainly caused by the increase of flow-related erosion, such as rill erosion and piping erosion, in clay soils (Roering et al., 1999).

10.1002/2017WR020806

AGU Water Resources Research

Figure 4. Scatterplots of observed sediment yield versus variables related to catchment scale (CA), topography (SLP, RSLP, PGA, and Frag), and soil (Clay, Silt, Sand, and SOC). Black lines show the linear regression fit which is based on all available catchments. Blue and red dots represent data from catchments with and without available Q_{peak} data, respectively.

The relationship between VarSM and SY shown in Figure 5 is not consistent with some plot-scale experiments (Bryan, 2000). Because soil water content may increase soil resistance to rain splash and concentrated flow (Gilley et al., 1993), intense rainfall occurring days before greening of the land in the arid and semiarid tropics can produce high SY (Nachtergaele et al., 2001). Thus, it is expected that the seasonality of soil moisture (expressed as VarSM) should be positively correlated with SY. But as presented in Figure 5 and Tables 2 and 3, VarSM does not have any positive impacts on the spatial and temporal variability of SY. It should also

10.1002/2017WR020806

Figure 5. Scatterplots of observed sediment yield versus land use (Wood, Grass, and Crop), population (POP), lithology (LITH), moisture (VarSM), and climate (P, Q, and Q_{peak}). Black lines show the linear regression fit which is based on all available catchments. Blue and red dots represent data from catchments with and without available Q_{peak} data, respectively.

be noted that VarSM is not necessarily a proxy for rainfall intensity and can also be very low in some highrelief terrain, such as the mountainous western U.S. (supporting information Figure S3). Because high-relief terrain usually corresponds to high SY (supporting information Figure S1), it is possible that any positive impacts of VarSM on SY have been cancelled out by the negative correlations between VarSM and topography variables (supporting information Table S1).

Figure 6. Principal component analysis (PCA) with environmental factors as active variables and sediment yield as supplementary variables. "a," "b," "c," and "d" represent sediment yield from catchments in the size range of 0.1-10, 10-30, 30-100, and 100-200 km², respectively. Vectors in red are the loadings of each environmental factor for the first and second PCs. The projection of each vector onto the PCs is a measure of the correlation between the corresponding factor and the PCs. The projection of each blue dot onto the PCs is a measure of the correlation between the PCs.

Catchment scale (expressed as CA) is known to have two opposite effects on SY: in catchments dominated by steep topography, SY could increase with CA due to the effect of larger concentrated water flow, but in catchments where flat topography dominates, SY could decrease with CA due to the effect of larger sediment deposition (de Vente & Poesen, 2005; Foster, 1982; Pelletier, 2012; Poesen et al., 1996). In Figures 4 and 6, CA is negatively correlated with SY and on the opposite side of the four SY scale groups ("a," "b," "c," and "d") in the second PC. These results might indicate the importance of representing sediment deposition in SY modeling in ESMs. However, we are cautious about the explanation because the relationship between CA and SY on the second PC may also be caused by the negative relationship of CA with Q and P (Figure 6).

Several factors including SOC and LITH that were previously known to significantly affect SY in plot or field experiments are not strongly correlated to SY in our analyses (Hassan et al., 2008; Rapp, 1998). Many plot-scale and catchment-scale soil erosion models include SOC as a negative factor to soil detachment (Panagos et al., 2014) because SOC can increase soil aggregation and cohesion (Rapp, 1998). As shown in both Figure 4 and Table 2, the inclusion of SOC for SY modeling in ESMs might not reduce simulation uncertainty. A possible reason is the lack of high-quality fine-scale SOC data to allow incorporation of the SOC effect on SY at the global scale. High erosion rates were documented previously in loess-rich and other regions where substrate strength is weak (Dadson et al., 2003; Syvitski & Milliman, 2007), such as the Loess Plateau of China (Hassan et al., 2008). In this study, we do not find strong correlations between lithology (expressed as LITH) and the spatial variability of SY (Figure 5 and supporting information Figure S3), even though many of the study catchments are located in loess-rich regions. One possible reason is the poor quality of the lithol-

ogy score data. As shown in Figure 5, the lithology score only varies in the range of 0–3 across the globe in our data set even though rock strength may vary over three orders of magnitude in relatively small regions such as the island of Taiwan (Dadson et al., 2003). In addition, as described by Vanmaercke et al. (2014b), the interaction between lithology and other stronger erosion factors (supporting information Table S1), i.e., topography and land use, may also weaken its main effect on SY.

Importantly, it should be noted that the dependence of SY on the topography and climate PCs in Figure 6 is almost consistent for the four SY scale groups. This might suggest that the correlations between SY and the environmental factors are not significantly dependent on catchment scale in the size range of 0.1–200 km². This provides some support for the statistically significant relationships identified in this study to provide a starting point for modeling SY in ESMs.

3.2. Effects of Land Cover and Management on SY

Anthropogenic factors are also known to be important for SY. To examine the global-scale effects of land use and management practices on SY, we used the one-way ANOVA test to evaluate the difference of SY in three land use and management groups: noncropland, traditional cropland, and cropland with soil conservation management (Figure 7). This test rejected the hypothesis that the three groups have statistically equal mean SYs ($p = 9.6 \times 10^{-5}$). In comparison with "pristine" catchments (mean SY: 391 t km⁻² yr⁻¹), cropland-dominated catchments have significantly larger SY (mean: 569 t km⁻² yr⁻¹; p < 0.05), but cropland with soil conservation management (e.g., reduced tillage, no tillage, and stone lines) has insignificantly smaller SY (mean: 213 t km⁻² yr⁻¹; p > 0.05).

The above results are consistent with the findings of previous studies that human activities can affect SY greatly (Lanckriet et al., 2012; Maetens et al., 2012; Montgomery, 2007; Nyssen et al., 2009; Vanmaercke et al., 2015). First, the development of ephemeral gully on cropland could be significantly faster than that

Table 2

Partial Correlation Coefficient (Partial r) and the Corresponding p-Value for the Mean Annual Value of Each Controlling Variable in Table 1 With the Natural Logarithm of Mean Annual Sediment Yield, In(SY)

Variable	Factor	Partial <i>r</i>	<i>p</i> -Value
In(Q _{peak})	Climate	0.52	$4.9 imes10^{-29}$
ln(Q)	Climate	0.32	$7.5 imes 10^{-11}$
ln(PGA)	Topography	0.27	$2.2 imes10^{-8}$
ln(SLP)	Topography	0.23	$1.0 imes 10^{-5}$
ln(P)	Climate	0.20	$6.5 imes 10^{-5}$
Wood	Land use	-0.17	0.0012
Clay	Soil	0.15	0.0026
ln(RE)	Climate	0.13	0.0140

Note. Only variables that are significantly partially correlated (p < 0.05) are shown and variables in bold are very significantly partially correlated (p < 0.0001).

on dense woodland due to the decrease of critical flow shear stress τ_c (Nachtergaele et al., 2001). This may help explain the strikingly higher SY of 569 t km⁻² yr⁻¹ in traditional cropland compared to the mean erosion rate of 90 t km⁻² yr⁻¹ (Cerdan et al., 2010; de Vente & Poesen, 2005) reported for fine-scale plots and fields where sediment mobilization is dominated by rain splash, interrill, and rill erosion. Second, because the τ_c values of no-till soils could be twice as large as the τ_c of tilled soils due to topsoil compaction (Franti et al., 1999; Laflen & Beasly, 1960), soil conservation management, especially no tillage, can clearly decrease the erodibility of topsoils to concentrated flow compared with conventional tillage (Knapen et al., 2007). Overall, the mean SY of conserved cropland is much lower than that of traditional cropland (p < 0.05).

It should be noted that in the one-way ANOVA, anthropogenic influence is narrowly defined as agricultural activities including soil and water conservation measures. In fact, other human activities could

also increase or decrease SY significantly. For example, the occurrence of fire is often related to human activities and SY was reported to increase significantly after fire (Malmon et al., 2007; Scott et al., 1998; Warrick et al., 2012). By using available global fire data sets (Giglio et al., 2013; Randerson et al., 2012), it is possible to investigate the effect of fire in the future. In addition, as population (expressed as POP) is strongly correlated to SY in the linear regression shown in Figure 5, it is possible that other human activities, e.g., land leveling, logging and mining, are also important in increasing SY (Douglas, 1996; Maetens et al., 2012).

3.3. Effects of Environmental Factors on CY

When only the main effect is considered using simple linear regressions (Figures 8 and 9), factors including Silt, SOC, Wood, Grass, P, and SY show significant spatial correlations (*p*-value < 0.05) with the natural logarithm of the observed CY. More specifically, ln(CY) is positively correlated with Grass, P, and SY and negatively correlated with Silt, SOC, and Wood. However, the correlations of CY with Silt, SOC, Wood, Grass, and P (especially the negative correlations of CY with SOC) could be misleading because they may result from interactions of those factors with SY. To remove the possible interactions, simple linear regressions were also performed for POC concentration in suspended sediment (CY divided by SY) with the environmental factors (supporting information Figures S4 and S5). The new regressions clearly indicate that POC concentration in suspended sediment is not significantly correlated with vegetation productivity related factors (e.g., Wood and SOC), which is consistent with the finding of Galy et al. (2015). These analyses suggest that sediment yield is likely the primary control on POC transport efficiency (Galy et al., 2015).

The decreasing CY at high SY is often explained by the observed decrease in biospheric POC concentration with depth in the soil profiles (Galy et al., 2015; Jobbágy & Jackson, 2000): at low SY surface matrial characterized by high POC concentrations is preferentially detached and exported while at high SY more material in the deeper soil layers with lower POC concentrations is detached and exported. In addition, this relationship

Table 3

Partial Correlation Coefficient (Partial r) and the Corresponding p-Value for the Time Series of Each Climate Variable (Table 1) With the Natural Logarithm of Sediment Yield, In(SY)

Variable	Factor	Partial r	<i>p</i> -Value
In(Q _{peak})	Climate	0.52	$1.3 imes 10^{-101}$
In(Q)	Climate	0.43	$2.8 imes 10^{-68}$
ln(P)	Climate	0.33	$8.8 imes 10^{-38}$
In(RE)	Climate	0.23	$2.3 imes 10^{-18}$
VarSM	Moisture	-0.05	0.0535

Note. Variables in bold are very significantly partially correlated (p < 0.0001) and variable in italic are insignificant (p > 0.05).

can also be explained by the fact that water flow at high SY has much larger shear stress to carry higher fractions of coarse material such as sand and gravel that is characterized by low biospheric POC concentrations (Lal, 2003). A notable strength of the latter interpretation is that it can also explain the lack of negative correlation between POC concentration and SOC shown in supporting information Figure S4.

3.4. Event-Scale Variables

Many existing SY models simulate global and regional SY at the annual time step using annual rainfall and/or runoff as drivers (de Vente et al., 2013; de Vente & Poesen, 2005; Quinton et al., 2006; Tucker & Bras, 2000). However, studies indicated that most of the observed SY in small catchments may be contributed by several large storms (Beven et al., 2005; Dadson et al., 2003; Gonzalez-Hidalgo et al.,

2010). To examine whether SY in small catchments should be modeled at the event-scale, we compared the SY dependence on the two event-scale variables (peak runoff Q_{peak} and rainfall erosivity RE) with the dependence on the two annual-scale variables (annual runoff Q and annual rainfall P) in the partial *r* and MLR analyses. It should be noted that only 472 catchments in our data set have Q_{peak} , RE and all other variables available, and most of these catchments are located in the U.S., Canada, and Puerto Rico. Although the scatterplots in Figures 4 and 5 show that the controlling factors of SY in these catchments do not differ significantly from those in the other catchments, we should be cautious to apply the conclusions drawn from the partial *r* and MLR analyses to other areas. As shown in Tables 2 and 3, Q_{peak} has larger partial *r* coefficients than other climate variables, such as Q, P, and RE. It means that in the study catchments Q_{peak} is more capable of explaining the spatial and temporal variability of SY than other climatic factors. This is consistent with the simple linear regression (Figure 5), in which Q_{peak} shows the largest main effect on the spatial variability of SY.

The results from our event-scale analysis are particularly interesting, given that Q_{peak} represents only the maximum daily runoff in a year and there are broad ranges of climate, topography, land cover, and other variations even in the 472 catchments. There could be two reasons for the explanatory power of Q_{peak} : (1) in most of the catchments where Q_{peak} has dominant effects, SY is overwhelmingly produced by storm-driven erosion (Dadson et al., 2003) or (2) eroded sediment deposited in the channels or lowland (Nyssen et al., 2004) can be more effectively transported downstream during large rain events when sediment transport capacity significantly increases (Dadson et al., 2003; Foster, 1982). In the first case, large storms may trigger gully erosion in low-relief terrain (Palleiro et al., 2014; Poesen et al., 1996, 2003) and landsliding in high-relief terrain (Dadson et al., 2003). In the second case, large storms may trigger debris flow that flushes colluvial sediment from the mountain belt (Dadson et al., 2003).

RE is used in the well-known Revised Universal Soil Loss Equation (RUSLE) to represent the effect of interrill erosion on soil detachment rates (Panagos et al., 2014; Renard et al., 1997). However, as presented in Tables 2 and 3, RE does not achieve comparable explanatory power to Q_{peak} . In fact RE is even worse than annual rainfall in explaining the spatial and temporal variability of the SY subset. It should be noted that the 3-hourly meteorological forcing data set used in the Variable Infiltration Capacity (VIC) model simulation may not be reliable in resolving the short-term variations of rainfall at the small catchment scale. This possibility indicates that SY models that use RE to represent rainfall-driven erosion may require high-quality

Figure 8. Scatterplots of observed POC yield (CY) versus variables related to catchment scale (CA), topography (SLP, RSLP, PGA, and Frag), and soil (Clay, Silt, Sand, and SOC). Black lines show the linear regression fits.

fine-resolution rainfall data to produce reasonable estimates of SY. In addition, the low contribution of interrill erosion to SY may also partly explain the poor performance of RE here (Poesen et al., 2003).

3.5. Regressions of SY and CY

As presented above, the relationships between SY and environmental factors can be very complex because of interactions among processes and environmental variables. By using an ensemble of stepwise backward MLR, we identified seven variables that are influential on the spatial variability of SY, including variables

 $ln(SY) = 1.44 - 0.01 Wood + 0.09 Clay - 0.03 Frag + 0.64 ln(Q_{peak}) + 0.49 ln(PGA).$ (1)

Figure 9. Scatterplots of observed POC yield (CY) versus land use (Wood, Grass, and Crop), population (POP), lithology (LITH), moisture (VarSM), and climate (P and Q), and sediment yield (SY). Black lines show the linear regression fits.

Figure 10. Observed sediment yield versus predicted sediment yield based on multiple linear regression (MLR) models. The shown formula is the MLR model used to predict sediment yield. The black dashed line is the 1:1 reference. The green solid line is the linear regression between the observed and predicted sediment yield.

The *p*-value of the *F*-statistics and the Akaike information criterion (AIC) score for the MLR are 3.68 imes 10⁻⁴⁵ and 1,612, respectively, indicating that this empirical relationship is significant. The high AIC score indicates that the regression models can achieve the best match with observations using the least number of parameters. The standard error (SE) of the intercept is 0.365 and the SEs of the coefficients for In(Q_{peak}), In(PGA), Wood, Clay, and Frag are 0.053, 0.064, 0.002, 0.012, and 0.011, respectively. The Wood and Frag terms in equation (1) are relatively less significant because of their high correlations with each other and also with ln(Q_{peak}) and ln(PGA) (see supporting information Table S1). Because cropland with and without soil conservation management has very different SY values, cropland was not selected as a predictor in equation (1). As discussed in section 3.3, however, human activities are an important factor in determining SY and their effects must be considered carefully based on crop management. In the MLR model, topsoil rock fragment (expressed as Frag) is found to have influence on the spatial variability of SY that is independent of the effects of topography. It is consistent with the findings that Frag increases soil resistance to both rainfall-driven and runoff-driven erosion (Poesen et al., 1994, 1999; van Wesemael et al., 1995). However, its effect is rarely included in empirical and process-based SY models. In equation (1), affected by the changes of climate, land use, and land cover, some terms (e.g., Wood) are possibly not stationary, which can reduce the effectiveness of the MLR as the best linear unbiased estimator. To reduce the impact of this nonstationarity, dynamic land use and land cover data could be useful in future studies.

By comparing Figure 10 with Figures 4 and 5, we can see that the predicted SY is mainly controlled by the spatial variability of Q_{peak} . Meanwhile, there is a visible bias shown in Figure 10 that is related to the overestimates of SY when Q_{peak} is in the range of 10–100 mm (supporting information Figure S7). By plotting the model-data residual as a function of the latitude and longitude (supporting information Figure S8), we notice that the catchments where SY is poorly predicted by equation (1) are mainly located around Lake Tahoe and in Wisconsin and Minnesota where a large amount of runoff is produced by snowmelt. It is possible that peak runoff fed by snowmelt water is less powerful in generating large SY (Puustinen et al., 2007) or frozen soil in those areas is less erodible (Renard et al., 1997).

The optimal MLR for CY was chosen by removing variables recursively with the *p*-value of their *F*-statistics larger than 0.05. This stepwise MLR process identified that CY is only significantly correlated with SY and not any other variables. The optimal regression of CY can be written as (Figure 11):

$$CY = 0.061 SY^{0.766}$$
, (2)

which is similar to the relationship reported for large river basins (Galy et al., 2015). When expressed as the natural logarithm of SY and CY, this equation is equivalent to $\ln(CY)=0.766 \ln(SY)-2.789$. The *p*-value of the *F*-statistics and the AIC score for the regression are 3.65×10^{-10} and 85, respectively, and the SEs of the intercept and the slope of $\ln(SY)$ are 0.423 and 0.079, respectively. As POC and sediment yield from bed load and large woody debris are hardly ever measured, equation (2) mainly reflects the relationship between CY and SY from suspended load. The strong correlation (r = 0.89) indicates that POC and sediment are physically bound in soils, so they are detached and exported through the same hydrological processes (McDowell et al., 1989; Smith et al., 2013). Further, as the exponent of SY in equation (2) is less than unity, the modeled POC concentration in exported sediment declines with the increase of SY in Figure 12 (Galy et al., 2015; Ludwig et al., 1996). As stated in section 3.3, this relationship is consistent with the findings of much lower POC export efficiency from high yield catchments, especially on steep topography (Kao & Liu, 1997; Kim et al., 2010; Lee et al., 2016; Lloret et al., 2013). As described above, the yield of POC and sediment from bed load and large woody debris may not follow the relationship of equation (2) but they could be important for small catchments.

Figure 11. Observed POC yield versus predicted POC yield. The shown formula is the regression model used to predict POC yield. The black dashed line is the 1:1 reference.

In Galy et al. (2015), similar relationships were established separately for both biospheric POC and petrogenic POC using measurements collected worldwide from 70 large river systems. We reanalyzed the data of Galy et al. (2015) for the sum of biospheric POC and petrogenic POC and established a similar relationship for the total POC data (supporting information Figure S9): $CY = 0.042 SY^{0.753}$. According to the river continuum theory (Vannote et al., 1980), the river mouth data of large watersheds may not reflect the characteristics of low-order streams. Indeed, when comparing the two regressions, POC concentrations measured in small catchments of our data set are higher than those measured in large river basins of Galy et al. (2015). This contrast is confirmed by the t test of POC concentrations in suspended sediment between our data set and Galy et al. (2015), as shown in Figure 12 (p-value = 0.0079). Possibly a large fraction of POC mobilized at the headwater is biogeochemically modified during transport and the loss cannot be compensated by carbon mobilization through bank and channel erosion (Bouchez et al., 2010; Ward et al., 2015).

4. Discussion

4.1. Implications for Representing SY in ESMs

Empirical and process-based SY models often have limited capability in simulating SY from small catchments at regional and continental scales (de Vente et al., 2013; de Vente & Poesen, 2005). Instead, these

models either focus on simulating SY from large river basins at coarse spatial resolutions (de Vente & Poesen, 2005; Pelletier, 2012; Syvitski & Milliman, 2007) or SY from single small catchments (Yu et al., 1999). Using data from over 1,000 small catchments around the world, our analyses focus on the spatial scales relevant to ESMs and suggest some directions to represent SY in ESMs.

First, to represent SY at a grid cell and in subgrid landunits of ESMs, it is not sufficient to simulate only rainsplash erosion, sheet flow erosion, and rill erosion as commonly represented in many existing SY models (de Vente & Poesen, 2005) because rainsplash erosion is unlikely a dominant sediment source at the scale of the grid and subgrid scale of ESMs and shallow overland flow and rill flow have limited sediment transport capacity to produce high SY due to their hydraulic geometry (Gilley et al., 1990; Hairsine & Rose, 1991, 1992a, 1992b). Instead, other more complex erosion processes should be represented. For example, since

high SY is linked with traditional agriculture and seismic activities, it is vital to simulate ephemeral gully erosion for cropland-dominated catchments and landslide for steep and tectonic active catchments in ESMs (de Vente et al., 2013). These erosion processes can be implemented separately by either process-based approaches (Arnone et al., 2016; Gordon et al., 2007) or simple empirical relationships (Posner & Georgakakos, 2015; Roering et al., 1999). Otherwise, all erosion processes driven by concentrated flow (e.g., rill flow, gully flow, and channel flow) might be represented using a single method with different hydraulic geometry settings (Morgan et al., 1984). Sediment yield is determined by the balance between soil erosion and sediment deposition. Besides erosion processes, the representation of sediment transport capacity is also needed. The strong correlation between SY and Q_{peak} indicates that most of the detached sediment is likely deposited in gullies and river channels for a year or longer until very strong floods flush the deposited sediment effectively (Dadson et al., 2003).

Second, it is important to model SY in ESMs at the event scale (daily or subdaily time steps) to represent catastrophic mass transport during episodic events. As both storm-driven erosion and sediment transport capacity increase exponentially with the increased intensity of concentrated flow (Bryan, 2000; Dadson et al., 2003; Hairsine & Rose, 1992a, 1992b), modeling SY at the annual or monthly scale in ESMs would cause large uncertainties. In addition, event-scale SY modeling will be more valuable for studying river biogeochemistry as previous works show that major hydrologic events (e.g., flooding) may drive carbon cycling in river networks (Raymond et al., 2016).

Third, as identified by the analyses, several environmental factors should be accounted for when modeling specific soil erosion and sediment transport processes in ESMs. For example, the intensity of seismic activities may be included as a negative factor for the calculation of soil cohesion that is usually a function of soil internal cohesion and root reinforced cohesion in process-based landslide modeling approaches (Arnone et al., 2016). As runoff-driven erosion is likely the dominant sediment source, it is important to carefully evaluate the effect of ground cover on erosion processes. As shown in our analyses, vegetation cover may be a major factor in calculating the impacts of ground cover on SY (Morgan et al., 1984). Other factors such as leaf area index and surface litter that were not analyzed in this study may be also considered (van Dijk & Bruijnzeel, 2001). Soil erosion from cropland catchments is sensitive to land management practices. By changing soil erodibility and ground cover, land management activities can reduce the severity of concentrated flow erosion (i.e., rill and gully erosion) on hillslopes (Knapen et al., 2007). Thus, as a first step, land management should be included in SY modeling to account for its effects on soil erodibility and ground cover in ESMs (Flanagan & Nearing, 1995). Similarly, the factor of topsoil rock fragment can also be included by adjusting the soil erodibility and ground cover (Flanagan & Nearing, 1995; Panagos et al., 2014). It should be noted that as erosion mainly occurs on hillslopes, it would be necessary to process the GIS data related to the above factors in the hillslope fraction of the grid and subgrid cells of ESMs.

Lastly, it may be necessary to remove snowmelt driven runoff from the calculation of SY in ESMs because the analyses of the MLR of SY suggest that this part of runoff is not as effective as rainfall-driven runoff in detaching the top soils.

4.2. Implications for Representing CY in ESMs

This study indicates that for small catchments, SY is the primary control on CY. The likely reason is that POC is physically bound with sediment and the detachment and transport of POC and sediment are driven by the same mechanisms. Similar to large river basins (Galy et al., 2015), an exponential relationship between SY and CY (equation (2)) was established by this study. This empirical relationship may serve as a good starting point for simulating CY in ESMs. For process-based approaches to represent CY in ESMs, our analyses suggest that it may be not sufficient to infer CY from SY by accounting only for the decrease in POC concentration with depth in the soil profiles (Dialynas et al., 2016). Instead, the difference of POC concentrations in soil particles and the dynamics of soil particle distribution in SY should be both considered.

4.3. Limitations and Future Work

The low quality of GIS data could be a large limitation for data analysis and modeling of SY and CY at the global scale (de Vente et al., 2013). For example, the weak correlation between substrate strength (expressed as LITH) and SY could be caused by the severe underestimation of the spatial variability of lithology in the data of Dürr et al. (2005). GIS data related to soil characteristics (e.g., soil texture and SOC) still

have large uncertainties despite the high resolutions because these products were produced using less reliable methods, such as limited sampling and extrapolation. Further, high-quality fine temporal-resolution rainfall and runoff data can be vital for event-scale SY modeling in ESMs. Due to the lack of GIS data, not all land management practices are included in our analyses. This lack of land management data can also affect the accuracy of SY modeling in ESMs, especially in regions with large fractions of cropland. Several ongoing global land degradation assessment projects including GLASOD, ASSOD, SOVEUR, and LADA may provide new opportunities for modeling the effect of land use in the future (de Vente & Poesen, 2005).

Several factors (e.g., occurrence of fire and the presence of lakes and reservoirs) that are not included here but were discussed by previous studies could also be important in representing SY (García-Ruiz et al., 2015; Renschler & Harbor, 2002; Warrick et al., 2012). The potential presence of (large and small) reservoirs in the catchments could also strongly influence SY (Renschler & Harbor, 2002), and aquatic systems such as lakes and wetlands could play important roles in processing of eroded organic carbon (Collins, 1996; de Vente et al., 2013). Missing of these factors could contribute to underestimation or overestimation of SY in SY models and data analysis, so they should be further explored in future studies.

As noted, there are several regions in the world (e.g., South Asia and South America) that are underrepresented by our SY data. It is possible that because of the underrepresentation, our explanations for some environmental factors could have large uncertainties. The difference between SY values from gauging stations and from reservoir surveys could be another source of uncertainties for the data analysis of SY. Because reservoir surveys typically result in much larger values (sometimes double) obtained in measurements (Vanmaercke et al., 2011, 2014b), our explanations of the effects of environmental factors on SY in the African and European catchments could be biased. Also, because most of the catchments with available Q_{peak} data are located in the U.S., Canada and Puerto Rico, we caution that the conclusions drawn from the partial *r* analysis and the SY MLR may not be completely applicable for other regions.

Our analyses of CY has provided some valuable information about model development and POC processing in rivers. However, because the analyses are based on far fewer catchments than those used in the analyses of SY, our conclusions can be less robust and characterized by larger uncertainties. In contrast to large river basins, the issue of limited data size may be much more severe for small upland catchments because they each represent only small areas of the land surface. As the dynamics of river POC is an important component of the global carbon cycle, this study motivates more efforts on measuring CY from small catchments across the globe and along the gradients of river corridors. Another drawback of the CY data is that POC in bed load and large woody debris is not included in the data.

Lastly, our study shows that a considerable fraction of the eroded POC could be biogeochemically modified in the rivers during transport. To fully understand the fate of leached POC in rivers, a comprehensive modeling suite including sediment flux, river routing and river biogeochemistry would be needed in future studies (Li et al., 2013, 2015; Pelletier et al., 2015).

5. Conclusions

Sediment and POC yield could have significant impacts on fluvial systems and the global carbon cycle, but they are not yet represented in most ESMs because soil erosion and sediment transport processes depend on the spatial scales and vary greatly in the spatial and temporal dimensions. Existing SY models and data analysis either focus on large river systems that have much coarser spatial scales than ESMs or specific catchments that cannot represent the global variability of SY. By analyzing SY and CY data from 1081 and 38 catchments, respectively, in the size range of 0.1–200 km² across the globe using multiple statistical analysis techniques, we investigated possible directions to advance SY and CY modeling in ESMs. Analyses of the environmental control on SY across a large number of catchments showed some relationships that are consistent with and some that are different from those reported in previous studies. But among the many factors examined, we identified several including seismicity, land forest cover, and human activity (land management) that display relationships with SY that are almost consistent with those previously reported. The robustness of the relationships for catchments across very different environments around the globe suggests that these factors are important for SY modeling at the global scale. Among them, seismicity and land management are rarely included in current catchment-scale SY models.

Our analyses also showed that high SY occurs in areas with intense traditional agriculture, seismicity and/or heavy storms. SY is also highly influenced by Q_{peak} . These highlight the limitation of simulating SY by representing interrill and rill erosion alone. Further, our analyses indicated a high capability of event-scale variables in explaining SY variability across a large number of catchments. This suggests that SY modeling in ESMs should be implemented at the event scale to reproduce the catastrophic mass transport during episodic events. The large influence of human activities, particularly cropland management, is well supported by our analyses and should be accounted by SY modeling in ESMs.

Last but not least, we found that SY is the primary control on CY. Our analyses established a statistically significant empirical relationship ($R^2 = 0.785$) to calculate the yield (units: t km⁻² yr⁻¹) of POC and sediment jointly. Expressed as CY = 0.061 SY^{0.766}, this relationship is similar to that established by Galy et al. (2015) for large river systems, except that the POC concentrations in small catchments are significantly higher, suggesting that it could be important to represent biogeochemitry and other processes for understanding carbon cycling in rivers. Derived using catchments resolvable by ESMs, our relationship could be used as a starting point to model POC fluxes in ESMs, given that SY is modeled with considerations of the natural and anthropogenic factors identified in this study.

Further reaserch is needed to address the weaknesses in our approach, which includes (1) a better representation of catchments in Asia, South America, and Australia; (2) a better recognization of the errors and uncertainties in SY and CY measurements, such as gauging station sampling versus reservoir surveys; (3) a coordinated effort of simultaneous SY and CY measurement in small catchments and along river corridors across the globe; and (4) the incorporation of more accurate GIS data and more human activities.

Appendix A: Database Acknowledgements

We acknowledge the following sources for providing the SY and CY data used in this study. Abaci and Papanicolaou (2009), Alberts et al. (1978), Alvera and García-Ruiz (2000), Balasch et al. (1992), Balasch et al. (2005), Bednarczyk and Madeyski (1996), Beylich et al. (2006), Beylich and Kneisel (2009), Beyer-Portner (1998), Boeglin et al. (2005), Bogardi et al. (1983), Bogen (1995, 1996), Boix-Fayos et al. (2008), Brański and Banasik (1996), Butcher et al. (1993), Cammeraat (2002), Casalí et al. (2008), Cerro et al. (2014), Chaplot and Poesen (2012), Collins (1991, 1996), Csáfordi et al. (2009), Cui et al. (2007), Dawson et al. (2002), Deasy et al. (2011), Descroix (1994), de Vente et al. (2005), Diaconu (1969), Dostál et al. (2006), Douglas et al. (1992), Esteves and Ludwig (2003), Foster and Lees (1999), Fouz et al. (2006), Froehlich and Walling (2005), García-Ruiz et al. (2008), Gergov (1996), Griffiths (1979), Gögüs and Yener (1997), Gomez et al. (2003), Grauso et al. (2008), Guyot et al. (1996), Habersack (1996), Haigh et al. (2004), Harrington and Harrington (2009), Hartmann et al. (2014), Hasholt (1983), Hejduk et al. (2006), Hicks et al. (2000), Hope et al. (1997), Huygens et al. (2000), IAHS (1974), Jacinthe et al. (2004), Jantawat (1985), Jaoshvili (2002), Jauffret and Vérague (1986), Krasa et al. (2005), Kostadinov and Markovic (1996), Labadz et al. (1991), Lamalle et al. (1989), Laubel et al. (1999), Lawler (1994), Lefrançois et al. (2007), Lemin et al. (1987), Lewis et al. (2012), Llovet et al. (1998), Loebis and Taryana (1988), López-Tarazóna et al. (2010), Lu and Higgitt (1999), Mano et al. (2006), Martin et al. (1997), Martínez-Carreras et al. (2010), Marttila and Kløve (2010), Mast (2013), McDowell et al. (1989), McDowell and Asbury (1994), Meybeck et al. (2003), Nadal-Romero et al. (2007), Oeurng et al. (2011), Olefeldt et al. (2013), Olive and Rieger (1986), Olivry and Hoorelbeck (1989), Øygarden et al. (2006), Palleiro et al. (2014), Pavanelli and Rigotti (2007), Pawson et al. (2008), Pepin et al. (2010), Peterjohn and Correll (1984), Petkovic et al. (1999), Pickup et al. (1981), Puustinen et al. (2007), Rãdoane and Rãdoane (2005), Rai and Sharma (1998), Rapp et al. (1972), Regüés and Gallart (2004), Rejman and Rodzik (2006), Restrepo et al. (2006), Rodríguez-Blanco et al. (2013), Sadeghi et al. (2013), Schröder and Theune (1984), Sharma and Rai (2004), Shibata et al. (2001), Sickman et al. (2007), Sine and Agneessens (1978), Sisák et al. (2007), Smith et al. (2013), Sogon et al. (1999), Steegen (2001), Stone and Saunderson (1996), Stutter et al. (2006), Soler et al. (2008), Tamburino et al. (1990), Thodsen et al. (2004), Torri et al. (2006), Tropeano (1991), Van Dijk and Kwaad (1996), Van Hoestenberghe et al. (2008), Van Rompaey et al. (2005), Verstraeten (2000), Verstraeten and Poesen (2001), Verstraten (1977), Vihermaa et al. (2016), Voet (1997), Vuorenmaa et al. (2002), and White (1996).

Acknowledgments

We are grateful to many researchers and institutions, especially Amanda Henck Schmidt at Oberlin College, USGS Water Resources, Environment and Climate Change Canada, and the Terrestrial Hydrology Research Group at Princeton, for making their sediment yield and ancillary data available for this study. This research is supported by the Office of Science of the U.S. Department of Energy as part of the Earth System Modeling program through the Energy Exascale Earth System Model (E3SM) project. The Pacific Northwest National Laboratory is operated by Battelle for the U.S. Department of Energy under contract DE-AC05-76RLO1830. J.H. was supported by the German Science Foundation DFG (Cluster of Excellence "CliSAP," EXC177, Universität Hamburg and DFG-project HA4472/6-1). The SY and CY data set can be accessed in the accompanying supporting information Data Set S1.

References

- Abaci, O., & Papanicolaou, A. N. (2009). Long-term effects of management practices on water-driven soil erosion in an intense agricultural sub-watershed: Monitoring and modeling. *Hydrological Processes*, 23, 2818–2837. https://doi.org/10.1002/hyp.7380
- Alberts, E. E., Schuman, G. E., & Burwell, R. E. (1978). Seasonal runoff losses of nitrogen and phosphorus from Missouri Valley Loess Watersheds. *Journal of Environmental Quality*, 7(2), 203–208.
- Alvera, B., & García-Ruiz, J. M. (2000). Variability of sediment yield from a high mountain catchment, Central Spanish Pyrenees. Arctic, Antarctic, and Alpine Research, 32(4), 478–484.
- Arnone, E., Caracciolo, D., Noto, L. V., Preti, F., & Bras, R. L. (2016). Modeling the hydrological and mechanical effect of roots on shallow landslides. Water Resources Research, 52, 8590–8612. https://doi.org/10.1002/2015WR018227
- Aufdenkampe, A. K., Mayorga, E., Raymond, P. A., Melack, J. M., Doney, S. C., Alin, S. R., ... Yoo, K. (2011). Riverine coupling of biogeochemical cycles between land, oceans, and atmosphere. *Frontiers in Ecology and the Environment*, *9*, 53–60. https://doi.org/10.1890/100014
- Balasch, J. C., Batalla, R. J., Poch, R. M., & Vericat, D. (2005). Patterns of suspended sediment transport in two forested Mediterranean mountainous basins, Ribera Salada, Catalan Pre-Pyrenees, NE Spain. *IAHS Publication*, 299, 131–138.
 - Balasch, J. C., Castelltort, X., Llorens, P., & Gallart, F. (1992). Hydrological and sediment dynamics network design in a Mediterranean mountainous area subject to gully erosion. IAHS Publication, 210, 423–432.
- Bednarczyk, T., & Madeyski, M. (1996). Erosional processes in small Carpathian watersheds. IAHS Publication, 236, 399-404.
- Beven, K., Heathwaite, L., Haygarth, P., Walling, D., Brazier, R., & Withers, P. (2005). On the concept of delivery of sediment and nutrients to stream channels. *Hydrological Processes*, 19, 551–556. https://doi.org/10.1002/hyp.5796
- Beyer-Portner, N. (1998). Erosion des bassins versants alpins par ruissellement de surface, Communication 6. In Laboratoire de Constructions Hydrauliques (340 pp.). Lausanne, Switzerland: Ecole Polytechnique Fédérale de Lausanne.
- Beylich, A. A., & Kneisel, C. (2009). Sediment budget and relief development in Hrafndalur, subarctic oceanic eastern Iceland. Arctic, Antarctic, and Alpine Research, 41, 3–17.
- Beylich, A. A., Sandberg, O., Molau, U., & Wache, S. (2006). Intensity and spatio-temporal variability of fluvial sediment transfers in an Arcticoceanic periglacial environment in northernmost Swedish Lapland, Latnjavagge catchment. Geomorphology, 80, 114–130.
- Boeglin, J., Probst, J., Ndam-Ngoupayou, J., Nyeck, B., Etcheber, H., Mortatti, J., & Braun, J. (2005). Soil carbon stock and river carbon fluxes in humid tropical environments: The Nyong River Basin (South Cameroon). In Soil erosion and carbon dynamics. Advances in Soil Science (pp. 275–288). Boca Raton, FL: CRC Press.
- Bogardi, I., Fogel, M., Duckstein, L., & Bardossy, A. (1983). Estimation of sediment loading into agricultural reservoirs with scarce data. IAHS Publication, 147, 87–98.
- Bogen, J. (1995). Sediment transport and deposition in mountain rivers. In I. Foster, A. Gurnell, & B. Webb (Eds.), Sediment and water quality in river catchments (pp. 437–451). Chichester, UK: John Wiley.
- Bogen, J. (1996). Erosion and sediment yield in Norwegian rivers. IAHS Publication, 236, 73-84.
- Boix-Fayos, C., de Vente, J., Martínez-Mena, M., Barberá, G. G., & Castillo, V. (2008). The impact of land use change and check-dams on catchment sediment yield. *Hydrological Processes*, 22, 4922–4935. https://doi.org/10.1002/hyp.7115
- Bork, H. R., Li, Y., Zhao, Y., Zhang, J., & Shiquan, Y. (2001). Land use changes and gully development in the upper Yangtze River Basin, SW-China. *Journal of Mountain Science*, 19, 97–103.
- Bouchez, J., Beyssac, O., Galy, V., Gaillardet, J., France-Lanord, C., Maurice, L., & Moreira-Turcq, P. (2010). Oxidation of petrogenic organic carbon in the Amazon floodplain as a source of atmospheric CO₂. *Geology*, *38*(3), 255–258. https://doi.org/10.1130/G30608.1
- Brański, J., & Banasik, K. (1996). Sediment yields and denudation rates in Poland. IAHS Publication, 236, 133–138.
- Broeckx, J., Vanmaercke, M., Bălteanu, D., Chendeş, V., Sima, M., Enciu, P., & Poesen, J. (2016). Linking landslide susceptibility to sediment yield at regional scale: Application to Romania. *Geomorphology*, *268*, 222–232.
- Bryan, R. B. (2000). Soil erodibility and processes of water erosion on hillslope. Geomorphology, 32, 385–415. https://doi.org/10.1016/S0169-555X(99)00105-1
- Butcher, D. P., Labadz, J. C., Potter, A. W. R., & White, P. (1993). Reservoir sedimentation rates in the Southern Pennine region, UK. In
- J. McManus & R.W. Duck (Eds.), Geomorphology and sedimentology of lakes and reservoirs (pp. 73–93). Chichester, UK: John Wiley. Cammeraat, L. H. (2002). A review of two strongly contrasting geomorphological systems within the context of scale. *Earth Surface Processes and Landforms*, *27*, 1201–1222. https://doi.org/10.1002/esp.421
- Casalí, J., Gastesi, R., Álvarez-Mozos, J., De Santisteban, L. M., Del Valle deLersundi, J., Giménez, R., . . . Donézar, M. (2008). Runoff, erosion, and water quality of agricultural watersheds in central Navarre (Spain). *Agricultural Water Management*, 95, 1111–1128.
- Center for International Earth Science Information Network Columbia University, & Centro Internacional de Agricultura Tropical. (2005). Gridded Population of the World, Version 3 (GPWv3): Population density grid. Palisades, NY: NASA Socioeconomic Data and Applications Center. Retrieved from https://doi.org/10.7927/H4XK8CG2
- Cerdan, O., Govers, G., Le Bissonnais, Y., Van Oost, K., Poesen, J., Saby, N., . . . Dostal, T. (2010). Rates and spatial variations of soil erosion in Europe: A study based on erosion plot data. *Geomorphology*, *122*, 167–177. https://doi.org/10.1016/j.geomorph.2010.06.011
- Cerro, I., Sanchez-Perez, J. M., Ruiz-Romera, E., & Antigüedad, I. (2014). Variability of particulate (SS, POC) and dissolved (DOC, NO₃) matter during storm events in the Alegria agricultural watershed. *Hydrological Processes*, *28*, 2855–2867. https://doi.org/10.1002/hyp.9850 Chaplot, V., & Poesen, J. (2012). Sediment, soil organic carbon and runoff delivery at various spatial scales. *Catena*, *88*, 46–56.
- Cohen, S., Willgoose, G., & Hancock, G. (2009). The mARM spatially distributed soil evolution model: A computationally efficient modeling framework and analysis of hillslope soil surface organization. *Journal of Geophysical Research*, *114*, F03001. https://doi.org/10.1029/
 - 2008JF001214
- Cole, J. J., Prairie, Y. T., Caraco, N. F., McDowell, W. H., Tranvik, L. J., Striegl, R. G., . . . Melack, J. (2007). Plumbing the global carbon cycle: Integrating inland waters into the terrestrial carbon budget. *Ecosystems*, *10*, 172–185. https://doi.org/10.1007/s10021-006-9013-8
 Collins, D. N. (1991). Climatic and glaciological influences on suspended sediment transport from an Alpine glacier. *IAHS Publication*, *203*, 3–12.
 Collins, D. N. (1996). Sediment transport from glacierized basins in the Karakoram Mountains. *IAHS Publication*, *236*, 85–96.
- Coulthard, T. J., Kirkby, M. J., & Macklin, M. G. (1996). A cellular automation landscape evolution model. In R. J. Abrahart (Ed.), Proceedings of the first international conference on GeoComputation (Vol. 1, pp. 248–281). Leeds, UK: School of Geography, University of Leeds.
- Csáfordi, P., Kalicz, P., Gribovszki, Z., & Kucsara, M. (2009). Examination of dam induced sedimentation of small reservoir near Brennbergbánya. EGU General Assembly Conference, 11, EGU2009–7409-1.
- CTIC. (2008). Crop residue management survey. West Lafayette, IN: Author. Retrieved from http://www.ctic.org/resourcedisplay/255/ Cui, L., Li, Z., Guo, Y., & Zhu, Y. (2007). Fractal-information-dimension-based relationship between sediment yield and topographic feature of watershed. Acta Pedologica Sinica, 44(2), 197–203.

Dadson, S. J., Hovius, N., Chen, H., Brian Dade, W., Hsieh, M.-L., Willett, S. D., . . . Lin, J.-C. (2003). Links between erosion, runoff variability and seismicity in the Taiwan orogen. *Nature*, 426, 648–651.

- Dawson, J. J. C., Billet, M., Neal, F. C., & Hill, S. (2002). A comparison of particulate, dissolved and gaseous carbon in two contrasting upland streams in the UK. Journal of Hydrology, 257, 226–246.
- Deasy, C., Baxendale, S. A., Heathwaite, A. L., Ridall, G., Hodgkinson, R., & Brazier, R. E. (2011). Advancing understanding of runoff and sediment transfers in agricultural catchments through simultaneous observations across scales. *Earth Surface Processes and Landforms*, 36(13), 1749–1760.
- Descroix, L. (1994). L' Erosion actuelle dans la partie occidentale des Alpes du Sud (PhD thesis, 337 pp.). Lyon, France: l'Université Lumiére Lyon II.
- de Vente, J., & Poesen, J. (2005). Predicting soil erosion and sediment yield at the basin scale: Scale issues and semi-quantitative models. *Earth-Science Reviews*, 71, 95–125.
- de Vente, J., Poesen, J., & Verstraeten, G. (2005). The application of semi-quantitative methods and reservoir sedimentation rates for the prediction of basin sediment yield in Spain. *Journal of Hydrology*, *305*, 63–86.
- de Vente, J., Poesen, J., Verstraeten, G., Govers, G., Vanmaercke, M., Van Rompaey, A., . . . Boix-Fayos, C. (2013). Predicting soil erosion and sediment yield at regional scales: Where do we stand? *Earth-Science Reviews*, *127*, 16–29. https://doi.org/10.1016/j.earscirev.2013.08.014

Diaconu, C. (1969). Résultats de l'étude de l'écoulement des alluvions en suspension des rivières de la Roumanie. Bulletin of the International Association of Scientific Hydrology, 14, 51–89.

Dialynas, Y. G., Bastola, S., Bras, R. L., Billings, S. A., Markewitz, D., & Richter, D. de B. (2016). Topographic variability and the influence of soil erosion on the carbon cycle. *Global Biogeochemical Cycles*, 30, 644–660. https://doi.org/10.1002/2015GB005302

Dostál, T., Janecek, M., Kliment, Z., Krása, J., Langhammer, J., Váška, J., & Vrana, K. (2006). Czech Republic. In J. Boardman & J. Poesen (Eds.), Soil erosion in Europe (pp. 107–116). Chichester, UK: John Wiley.

- Douglas, I. (1996). The impact of land-use changes, especially logging, shifting cultivation, mining and urbanization on sediment yields in tropical Southeast Asia: A review with special reference to Borneo. *IAHS Publication, 236*, 463–472.
- Douglas, I., Spencer, T., Greer, T., Bidin, K., Sinun, W., & Meng, W. W. (1992). The impact of selective commercial logging on stream hydrology, chemistry and sediment loads in the Ulu Segama rain forest, Sabah, Malaysia. *Philosophical Transactions of the Royal Society B*, 335, 397–406.
- Dunne, T., Malmon, D. V., & Dunne, K. B. J. (2016). Limits on the morphogenetic role of rain splash transport in hillslope evolution. Journal of Geophysical Research: Earth Surface, 121, 609–622. https://doi.org/10.1002/2015JF003737
- Dürr, H. H., Meybeck, M., & Dürr, S. H. (2005). Lithologic composition of the Earth's continental surfaces derived from a new digital map emphasizing riverine material transfer. *Global Biogeochemical Cycles*, *19*, GB4S10. https://doi.org/10.1029/2005GB002515

Esteves, J. G., & Ludwig, W. (2003). Transfert de matière en suspension et de carbone particulaire dans le bassin versant de la Têt, Sud de la France. IAHS Publication, 278, 442–447.

Flanagan, D. C., & Nearing, M. (1995). USDA–Water Erosion Prediction Project hillslope profile and watershed model documentation (Rep. 10). West Lafayette, IN: National Soil Erosion Research Laboratory.

Foster, G. R. (1982). Modeling the erosion process. In C. T. Haan (Ed.), *Hydrologic modeling of small watersheds* (pp. 295–380). St. Joseph, Michigan: American Society of Agricultural and Biological Engineers.

Foster, I. D. L., & Lees, J. A. (1999). Changing headwater suspended sediment yields in the LOIS catchments over the last century: A paleolimnological approach. *Hydrological Processes*, 13, 1137–1153.

- Fouz, P. S., Vázquez, E. V., & González, A. P. (2006). Heavy-metal losses from an agroforestry catchment. Communications in Soil Science and Plant Analysis, 37, 15–20.
- Franti, T. G., Laflen, J. M., & Watson, D. A. (1999). Predicting soil detachment from high-discharge concentrated flow. *Transactions of the* ASAE, 42(2), 329–335.

Froehlich, W., & Walling, D. (2005). Using environmental radionuclides to elucidate sediment sources within a small drainage basin in the Polish Flysch Carpathians. *IAHS Publication*, 291, 102–112.

Galy, V., Peucker-Ehrenbrink, B., & Eglinton, T. (2015). Global carbon export from the terrestrial biosphere controlled by erosion. *Nature*, *521*, 204–207. https://doi.org/10.1038/nature14400

García-Ruiz, J. M., Beguería, S., Nadal-Romero, E., González-Hidalgo, J. C., Lana-Renault, N., & Sanjuán, Y. (2015). A meta-analysis of soil erosion rates across the world. Geomorphology, 239, 160–173. https://doi.org/10.1016/j.geomorph.2015.03.008

García-Ruiz, J. M., Regüés, D., Alvera, B., Lana-Renault, N., Serrano-Muela, P., Nadal-Romero, E., . . . Arnáez, J. (2008). Flood generation and sediment transport in experimental catchments affected by land use changes in the central Pyrenees. *Journal of Hydrology*, 356, 245–260. Gergov, G. (1996). Suspended sediment load of Bulgarian rivers. *GeoJournal*, 40, 387–396.

Giardini, D., Grunthal, G., Shedlock, K. M., & Zhang, P. Z. (1999). The GSHAP global seismic hazard map. Annals of Geophysics, 42(6), 1225–1230.
Giglio, L., Randerson, J. T., & van der Werf, G. R. (2013). Analysis of daily, monthly, and annual burned area using the fourth-generation global fire emissions database (GFED4). Journal of Geophysical Research: Biogeosciences, 118, 317–328. https://doi.org/10.1002/jgrg.20042

Gilley, J. E., Elliot, W. J., Laflen, J. M., & Simanton, J. R. (1993). Critical shear stress and critical flow rates for initiation of rilling. Journal of Hydrology, 142, 251–271. https://doi.org/10.1016/0022-1694(93)90013-Y

Gilley, J. E., Kottwitz, E. R., & Simanton, J. R. (1990). Hydraulic characteristics of rills. Transactions of the ASAE, 33, 1900–1906.

Gögüs, M., & Yener, A. (1997). Estimation of sediment yield rates of reservoirs in Turkey. In Proceedings of the Dix-neuvième Congrès des Grands Barrages (pp. 1265–1276). Florence, Italy: Commision Internationale des Grands Barrages.

Gomez, B., Trustrum, N. A., Hicks, D. M., Rogers, K. M., Page, M. J., & Tate, K. R. (2003). Production, storage, and output of particulate organic carbon: Waipaoa River basin, New Zealand. Water Resources Research, 39(6), 1161. https://doi.org/10.1029/2002WR001619

Gonzalez-Hidalgo, J. C., Batalla, R. J., Cerdà, A., & de Luis, M. (2010). Contribution of the largest events to suspended sediment transport across the USA. *Land Degradation & Development*, *21*, 83–91. https://doi.org/10.1002/ldr.897

Gordon, L. M., Bennett, S. J., Bingner, R. L., Theurer, F. D., & Alonso, C. V. (2007). Simulating ephemeral gully erosion in AnnAGNPS. Transactions of the ASABE, 50(3), 857–866.

Grauso, S., Pagano, A., Fattoruso, G., De Bonis, P., Onori, F., Regina, P., & Tebano, C. (2008). Relations between climatic-geomorphological parameters and sediment yield in a Mediterranean semi-arid area, Sicily, southern Italy. *Environmental Geology*, 54, 219–234.

Griffiths, G. A. (1979). High sediment yields from major rivers of the Western Southern Alps, New Zealand. *Nature*, 282, 61–63. Griffiths, G. A. (1981). Some suspended sediment yields from South Island catchments, New Zealand. *Water Resources Bulletin*, 17, 662–671.

Guyot, J. L., Filizola, N., Quintanilla, J., & Cortez, J. (1996). Dissolved solids and suspended sediment yields in the Rio Madeira basin, from the Bolivian Andes to the Amazon. *IAHS Publication*, 236, 55–63.

Habersack, H. M. (1996). Lack and surplus of sediments being transported by river systems. IAHS Publication, 236, 565–573.

Haigh, M. J., Jansky, L., & Hellin, J. (2004). Headwater deforestation: A challenge for environmental management. *Global Environmental Change*, 14, 51–61.

Hairsine, P. B., & Rose, C. W. (1991). Rainfall detachment and deposition: Sediment transport in the absence of flow-driven processes. Soil Science Society of America Journal, 552, 320–324.

Hairsine, P. B., & Rose, C. W. (1992a). Modeling water erosion due to overland flow using physical principles: 1. Sheet flow. *Water Resources Research*, 28, 237–243.

Hairsine, P. B., & Rose, C. W. (1992b). Modeling water erosion due to overland flow using physical principles: 2. Rill flow. Water Resources Research, 28(1), 245–250.

Harrington, S., & Harrington, J. (2009). The particulate fraction of nutrients transported in river loads in the South Western River Basin District, SWRBD, Ireland. EGU General Assembly, 11, EGU2009–7409-1.

Hartmann, J., Lauerwald, R., & Moosdorf, N. (2014). A brief overview of the GLObal River CHemistry Database, GLORICH. Procedia Earth and Planetary Science, 10, 23–27.

Hasholt, B. (1983). Dissolved and particulate load in Danish water courses. IAHS Publication, 141, 255-264.

Hassan, M. A., Church, M., Xu, J., & Yan, Y. (2008). Spatial and temporal variation of sediment yield in the landscape: Example of Huanghe (Yellow River). *Geophysical Research Letters*, 35, L06401. https://doi.org/10.1029/2008GL033428

Hejduk, L., Hejduk, A., & Banasik, K. (2006). Suspended sediment transport during rainfall and snowmelt–rainfall floods in a small lowland catchment, central Poland. In P. Owens & A. Collins (Eds.), Soil erosion and sediment redistribution in river catchments (pp. 94–100). Wallingford, UK: CAB International.

Henck, A., Huntington, K., Stone, J. O., Montgomery, D. R., & Hallet, B. (2011). Spatial controls on erosion in the Three Rivers region, western China. Earth and Planetary Science Letters, 303, 71–83. https://doi.org/10.1016/j.epsl.2010.12.038

Henck, A. H., Montgomery, D. R., Huntington, K. W., & Liang, C. (2010). Monsoon control of effective discharge, Yunnan and Tibet. Geology, 38, 975–978.

Hengl, T., de Jesus, J. M., MacMillan, R. A., Batjes, N. H., Heuvelink, G. B. M., Ribeiro, E., . . . Gonzalez, M. R. (2014). SoilGrids1km—Global soil information based on automated mapping. PLoS ONE, 9(8), e105992. https://doi.org/10.1371/journal.pone.0105992

Hicks, D. M., Gomez, B., & Trustrum, N. A. (2000). Erosion thresholds and suspended sediment yields, Waipaoa River Basin, New Zealand. Water Resources Research, 36(4), 1129–1142. https://doi.org/10.1029/1999WR900340

Hilton, R. G., Galy, V., Gaillardet, J., Dellinger, M., Bryant, C., O'Regan, M., ... Calmels, D. (2015). Erosion of organic carbon in the Arctic as a geological carbon dioxide sink. *Nature*, 524(7563), 84–87. https://doi.org/10.1038/nature14653

Hope, D., Billett, M. F., & Cresser, M. S. (1997). Exports of organic carbon in two river systems in NE Scotland. Journal of Hydrology, 193, 61-82.

Hovius, N. (1998). Controls on sediment supply by large rivers. In K. W. Shanley & P. J. McCabe (Eds.), *Relative role of eustasy, climate, and tectonism in continental rocks. Special Publications of SEPM Society for Sedimentary Geology* (Vol. 59, pp. 2–16). https://doi.org/10.2110/pec.98.59.0002

Huygens, M., Verhoeven, R., & de Sutter, R. (2000). Integrated river management of a small Flemish River Catchment. *IAHS Publication*, 263, 191–199.

IAHS. (1974). Gross sediment transport into the oceans (First Prelim. Ed. Tech. Rep. SC.74/WS/33), Paris, France: UNESCO.

Jacinthe, P.-A., Lal, R., Owens, L. B., & Hothem, D. L. (2004). Transport of labile carbon in runoff as affected by land use and rainfall characteristics. Soil & Tillage Research, 77, 11–23.

Jantawat, S. (1985). An overview of soil erosion and sedimentation in Thailand. In S. A. El-Swaify, W. C. Moldenhauer, & A. Lo (Eds.), Soil erosion and conservation (pp. 10–14). Ankeny, IA: Soil and Water Conservation of America Press.

Jaoshvili, S. (2002). The rivers of the Black Sea (EEA Tech. Rep. 71, 58 pp.). Copenhagen, Denmark: European Environmental Agency. Retrieved from http://www.reports.eea.europa.eu/technical_report_2002_71/en

Jauffret, D., & Vérague, J. (1986). Etude quantitative du transport en suspension de deux cours d'eau en milieu granitique tempéré océanique. In *Rapports scientifiques et techniques* (Vol. 3, 50 pp.). Caen, France: CNRS–Centrre de Géomorphologie.

Jobbágy, E. G., & Jackson, R. B. (2000). The vertical distribution of soil organic carbon and its relation to climate and vegetation. *Ecological* Applications, 10, 423–436.

Kao, S.-J., & Liu, K.-K. (1997). Fluxes of dissolved and nonfossil particulate organic carbon from an Oceania small river (Lanyang Hsi) in Taiwan. Biogeochemistry, 39, 255–269. https://doi.org/10.1023/A:1005864605382

Ke, Y., Leung, L. R., Huang, M., Coleman, A. M., Li, H., & Wigmosta, M. S. (2012). Development of high resolution land surface parameters for the Community Land Model. *Geoscientific Model Development*, 5(6), 1341–1362. https://doi.org/10.5194/gmd-5-1341-2012

Kempe, S. (1982). Long-term records of CO₂ pressure fluctuations in fresh water. In E. T. Degens (Ed.), *Transport of carbon and minerals in major world rivers* (Vol. 52, pp. 91–331). Hamburg, Federal Republic of Germany: Mitteilungen aus dem Geologisch-Paläontologischen Institut, University of Hamburg.

Kempe, S. (1984). Sinks of the anthropogenically enhanced carbon cycle in surface fresh waters. Journal of Geophysical Research, 89, 4657– 4676. https://doi.org/10.1029/JD089iD03p04657

Kim, S. J., Kim, J., & Kim, K. (2010). Organic carbon efflux from a deciduous forest catchment in Korea. *Biogeosciences*, 7, 1323–1334. https:// doi.org/10.5194/bg-7-1323-2010

Knapen, A., Poesen, J., Govers, G., Gyssels, G., & Nachtergaele, J. (2007). Resistance of soils to concentrated flow erosion: A review. Earth-Science Reviews, 80, 75–109. https://doi.org/10.1016/j.earscirev.2006.08.001

Krasa, J., Dostal, T., Van Rompaey, A., Vaska, J., & Vrana, K. (2005). Reservoirs' siltation measurements and sediment transport assessment in the Czech Republic, the Vrchlice catchment study. Catena, 64, 348–362.

Kostadinov, S., & Markovic, S. (1996). Soil erosion and effects of erosion control works in the torrential drainage basins of southeast Serbia. IAHS Publication, 236, 321–332.

Labadz, J. C., Burt, T. P., & Potter, A. W. R. (1991). Sediment yield and delivery in the blanket peat moorlands of the southern Pennines. Earth Surface Processes and Landforms, 16, 255–271. https://doi.org/10.1002/esp.3290160306

Lacoul, P., & Freedman, B. (2006). Environmental influences on aquatic plants in freshwater ecosystems. *Environmental Reviews*, 14, 89–136. https://doi.org/10.1139/a06-001

Laflen, J. M., & Beasly, R. P. (1960). Effects of compaction on critical tractive forces in cohesive soils (Res. Bull. 749). Columbia, MO: University of Missouri, Agricultural Experiment Station.

Lal, R. (2003). Soil erosion and the global carbon budget. Environment International, 29, 437–450. https://doi.org/10.1016/S0160-4120(02)00192-7

Lamalle, C., Petit, F., Koch, G., Hurtgen, C., & Pissart, A. (1989). Les transports en suspension et en solution dans la Burdinale, affluent de principale de la Méhaigne. Buletin de la Société Géographique de Liège, 25, 39–51.

Lanckriet, S., Araya, T., Cornelis, W., Verfaillie, E., Poesen, J., Govaerts, B., ... Nyssen, J. (2012). Impact of conservation agriculture on catchment runoff and soil loss under changing climate conditions in May Zeg-zeg (Ethiopia). Journal of Hydrology, 475, 336–349. https://doi. org/10.1016/j.jhydrol.2012.10.011

Laubel, A., Svendsen, L. M., Kronvang, B., & Larsen, S. E. (1999). Bank erosion in a Danish lowland stream system. Hydrobiologia, 410, 279–285.

Lawler, D. M. (1994). Recent changes in rates of suspended sediment transport in the Jökulsá á Sólheimasandi glacial river, southern Iceland. *IAHS Publication*, 224, 343–350.

Lee, M.-H., Payeur-Poirier, J.-L., Park, J.-H., & Matzner, E. (2016). Variability in runoff fluxes of dissolved and particulate carbon and nitrogen from two watersheds of different tree species during intense storm events. *Biogeosciences*, 13, 5421–5432. https://doi.org/10.5194/bg-13-5421-2016

Lefrançois, J., Grimaldi, C., Gascuel-Odoux, C., & Gilliet, N. (2007). Suspended sediment and discharge relationships to identify bank degradation as a main sediment source on small agricultural catchments. *Hydrological Processes*, 21, 2923–2933.

Lehner, B., Verdin, K., & Jarvis, A. (2008). New global hydrography derived from spaceborne elevation data. Eos, Transactions American Geophysical Union, 89(10), 93–94. https://doi.org/10.1029/2008EO100001

Lemin, G., Koch, G., Hurtgen, C., & Pissart, A. (1987). Les transports en suspension de la Meuse, l'Ourthe et la Höegne. Buletin de la Société Géographique de Liège, 22–23, 39–61.

Lewis, T., Lafrenière, M. J., & Lamoureux, S. F. (2012). Hydrochemical and sedimentary responses of paired High Arctic watersheds to unusual climate and permafrost disturbance, Cape Bounty, Melville Island, Canada. *Hydrological Processes*, 26(13), 2003–2018.

Li, H., Wigmosta, M. S., Wu, H., Huang, M., Ke, Y., Coleman, A. M., & Leung, L. R. (2013). A physically based runoff routing model for land surface and earth system models. *Journal of Hydrometeorology*, 14(3), 808–828. https://doi.org/10.1175/JHM-D-12-015.1

Li, H.-Y., Leung, L. R., Tesfa, T., Voisin, N., Hejazi, M., Liu, L., ... Yang, X. (2015). Modeling stream temperature in the Anthropocene: An earth system modeling approach. *Journal of Advances in Modeling Earth Systems*, 7, 1661–1679. https://doi.org/10.1002/2015MS000471

Li, Z., & Fang, H. (2016). Impacts of climate change on water erosion: A review. Earth-Science Reviews, 163, 94–117.

Lloret, E., Dessert, C., Pastor, L., Lajeunesse, E., Crispi, O., Gaillardet, J., & Benedetti, M. F. (2013). Dynamic of particulate and dissolved organic carbon in small volcanic mountainous tropical watersheds. *Chemical Geology*, 351, 229–244. https://doi.org/10.1016/j.chemgeo. 2013.05.023

Llovet, J., Bautista, S., Giovanardi, F., & Vallejo, V. (1998). Sediment production in burned catchments of eastern Spain. Paper presented at the proceedings of the 23th general assembly of EGU, HSC1–03. Nice, France.

Loebis, J., & Taryana, O. (1988). Erosion-sedimentation study for reservoir siltation control. Paper presented at the proceedings of the seminar on watershed research and management practices, ASEAN-US Watershed Project. Laguna, Philippines.

López-Tarazóna, J. A., Batalla, R. J., Vericat, D., & Balasch, J. C. (2010). Rainfall, runoff and sediment transport relations in a mesoscale mountainous catchment: The River Isábena (Ebro basin). Catena, 82(1), 23–34.

Lu, H., Koike, T., Fujii, H., Ohta, T., & Tamagawa, K. (2009). Development of a physically-based soil moisture retrieval algorithm for spaceborne passive microwave radiometers and its application to AMSR-E. Journal of the Remote Sensing Society of Japan, 29(1), 253–262.

Lu, X. X., & Higgitt, D. L. (1999). Sediment yield variability in the upper Yangtze, China. *Earth Surface Processes and Landforms*, 24, 1077–1093.

Ludwig, W., Probst, J. L., & Kempe, S. (1996). Predicting the oceanic input of organic carbon by continental erosion. *Global Biogeochemical Cycles*, 10, 23–41. https://doi.org/10.1029/95GB02925

Maetens, W., Vanmaercke, M., Poesen, J., Jankauskas, B., Jankauskiene, G., & Ionita, I. (2012). Effects of land use on annual runoff and soil loss in Europe and the Mediterranean: A meta-analysis of plot data. *Progress in Physical Geography*, 36, 599–653.

Malmon, D. V., Reneau, S. L., Katzman, D., Lavine, A., & Lyman J. (2007). Suspended sediment transport in an ephemeral stream following wildfire. Journal of Geophysical Research: Earth Surface, 112, F02006. https://doi.org/10.1029/2005JF000459

Mano, V., Moatar, F., Coynel, A., Etcheber, H., Ludwig, W., Meybeck, M., ... Schafer, J. (2006). Space and time variability of suspended particulate matter (SPM) transport in 32 French rivers (100 to 100,000 km2; daily to yearly). Paper presented at the sediment dynamics and the hydromorphology of fluvial systems ICCE IAHS international symposium. Dundee, Scotland.

Martin, C., Allée, P., Béguin, É., Kuzucuoglu, C., & Levant, M. (1997). Mesure de l'érosion mécanique des sols après un incendie de forêt dans le massif des Maures. *Géomorphologie*, *2*, 133–142.

Martin, E. E., Ingalls, A. E., Richey, J. E., Keil, R. G., Santos, G. M., Carlson, L. T., . . . Druffel, E. R. M. (2013). Age of riverine carbon suggests rapid export of terrestrial primary production in tropics. *Geophysical Research Letters*, 40, 5687–5691. https://doi.org/10.1002/2013GL057450

Martínez-Carreras, N., Udelhoven, T., Krein, A., Gallart, F., Iffly, J. F., Ziebel, J., . . . Walling, D. E. (2010). The use of sediment colour measured by diffuse reflectance spectrometry to determine sediment sources: Application to the Attert River catchment (Luxembourg). Journal of Hydrology, 382, 49–63.

Marttila, H., & Kløve, B. (2010). Dynamics of erosion and suspended sediment transport from drained peatland forestry. Journal of Hydrology, 388(3), 414–425.

Mast, M. A. (2013). Evaluation pf stream chemistry trends in US Geological Survey reference watersheds, 1970–2010. Environmental Monitoring and Assessment, 185, 9343–9359.

McDowell, L. L., Willis, G. H., & Murphree, C. E. (1989). Nitrogen and phosphorus yields in run-off from silty soils in the Mississippi Delta, USA. Agriculture, Ecosystems and Environment, 25, 119–137.

McDowell, W. H., & Asbury, C. E. (1994). Export of carbon, nitrogen and major ions from three tropical montane watersheds. *Limnology and Oceanography*, 39, 111–125.

Meade, R. H., Dunne, T., Richey, J. E., Santos, U. D. M., & Salati, E. (1985). Storage and remobilization of suspended sediment in the lower Amazon River of Brazil. Science, 228, 488–490.

Meybeck, M., Laroche, L., Dürr, H. H., & Syvitski, J. P. M. (2003). Global variability of daily total suspended solids and their fluxes in rivers. Global and Planetary Change, 39, 65–93.

Molnar, P., Anderson, R. S., & Anderson, S. P. (2007). Tectonics, fracturing of rock, and erosion. *Journal of Geophysical Research*, 112, F03014. https://doi.org/10.1029/2005JF000433

Montgomery, D. R. (2007). Soil erosion and agricultural sustainability. Proceedings of the National Academy of Sciences of the United States of America, 104, 13268–13272.

Morgan, R. P. C., Morgan, D. D. V., & Finney, H. J. (1984). A predictive model for the assessment of soil erosion risk. Journal of Agricultural Engineering Research, 30, 245–253.

Mutema, M., Chaplot, V., Jewitt, G., Chivenge, P., & Blöschl, G. (2015). Annual water, sediment, nutrient, and organic carbon fluxes in river basins: A global meta-analysis as a function of scale. Water Resources Research, 51, 8949–8972. https://doi.org/10.1002/2014WR016668 Nachtergaele, J., Poesen, J., Steegen, A., Takken, I., Beuselinck, L., Vandekerckhove, L., & Govers, G. (2001). The value of a physically based model versus an empirical approach in the prediction of ephemeral gully erosion for loess-derived soils. *Geomorphology*, 40, 237–252.

Nadal-Romero, E., Regüés, D., Latron, J., Lana-Renault, N., Serrano-Muela, P., & Martí-Bono, C. (2007). Badland areas: The main suspended sediment source in Central Spanish Pyrenees. Paper presented at the proceedings of the COST 634 international conference on off-site impacts of soil erosion and sediment transport. Prague, Czech Republic.

Nyssen, J., Clymans, W., Poesen, J., Vandecasteele, I., De Baets, S., Haregeweyn, N., ... Deckers, J. (2009). How soil conservation affects the catchment sediment budget—A comprehensive study in the north Ethiopian highlands. *Earth Surface Processes and Landforms*, 34, 1216–1233.

Nyssen, J., Poesen, J., Moeyersons, J., Deckers, J., Haile, M., & Lang, A. (2004). Human impact on the environment in the Ethiopian and Eritrean highlands—A state of the art. *Earth-Science Reviews*, 64, 273–320. 2004.

Oeurng, C., Sauvage, S., & Sánchez-Pérez, J. M. (2011). Assessment of hydrology, sediment and particulate organic carbon yield in a large agricultural catchment using the SWAT model. *Journal of Hydrology*, 401(3), 145–153.

Olefeldt, D., Roulet, N., Giesler, R., & Persson, A. (2013). Total waterborne carbon export and DOC composition from ten nested subarctic peatland catchments—Importance of peatland cover, groundwater influence, and inter-annual variability of precipitation patterns. *Hydrological Processes*, 27, 2280–2294. https://doi.org/10.1002/hyp.9358

Olive, L. J., & Rieger, W. A. (1986). Low Australian sediment yields—A question of inefficient sediment delivery. *IAHS Publication*, *159*, 305–322. Olivry, J. C., & Hoorelbeck, J. (1989). Erodibilité des terres noires de la vallée du Buëch, France, Alpes du Sud. *ORSTOM*, *25*, 95–110.

Øygarden, L., Lundekvam, H., Arnoldussen, A. H., & Børresen, T. (2006). Norway. In J. Boardman & J. Poesen (Eds.), Soil erosion in Europe (pp. 3–16). Chichester, UK: John Wiley.

Palleiro, L., Rodríguez-Blanco, M. L., Taboada-Castro, M. M., & Taboada-Castro, M. T. (2014). Hydroclimatic control of sediment and metal export from a rural catchment in northwestern Spain. *Hydrology and Earth System Sciences*, 18, 3663–3673. https://doi.org/10.5194/hess-18-3663-2014

Panagos, P., Meusburger, K., Ballabio, C., Borrelli, P., & Alewell, C. (2014). Soil erodibility in Europe: A high-resolution dataset based on LUCAS. *Science of the Total Environment*, 479–480, 189–200.

Pavanelli, D., & Rigotti, M. (2007). Relationship between suspended sediment concentration and total nitrogen, and comparison among basins for some Apennine Rivers. Paper presented at the proceedings of the COST 634 international conference on off-site impacts of soil erosion and sediment transport. Prague, Czech Republic.

Pawson, R. R., Lord, D. R., Evans, M. G., & Allott, T. E. H. (2008). Fluvial organic carbon flux from an eroding peatland catchment, southern Pennines, UK. Hydrology and Earth System Sciences, 12, 625–634.

Pelletier, J. D. (2012). A spatially distributed model for the long-term suspended sediment discharge and delivery ratio of drainage basins. Journal of Geophysical Research, 117, F02028. https://doi.org/10.1029/2011JF002129

Pelletier, J. D., Murray, A. B., Pierce, J. L., Bierman, P. R., Breshears, D. D., Crosby, B. T., ... Yager, E. M. (2015). Forecasting the response of Earth's surface to future climatic and land use changes: A review of methods and research needs. *Earth Future*, *3*, 220–251. https://doi. org/10.1002/2014EF000290

Pepin, E., Carretier, S., Guyot, J. L., & Escobar, F. (2010). Specific suspended sediment yields of the Andean rivers of Chile and their relationships to climate, slope, and vegetation. *Hydrological Sciences Journal*, 55(7), 1190–1205. https://doi.org/10.1080/02626667.2010.512868

Peterjohn, W. T., & Correll, D. L. (1984). Nutrient dynamics in an agricultural watershed: Observations on the role of a riparian forest. *Ecology*, 65(5), 1466–1475.

Petkovic, S., Dragovic, N., & Markovic, S. (1999). Erosion and sedimentation problems in Serbia. *Hydrological Sciences Journal*, 44, 63–77. Pickup, G., Higgins, R. J., & Warner, R. F. (1981). Erosion and sediment yield in the Fly River drainage basins, Papua New Guinea. In *Erosion*

and sediment transport in pacific rim steeplands (Vol. 132, pp. 438–456). International Association of Hydrological Sciences. Poesen, J., Boardman, J., Wilcox, B., & Valentin, C. (1996). Water erosion monitoring and experimentation for global change studies. *Journal*

of Soil and Water Conservation, 51(5), 386–390.

Poesen, J., de Luna, E., Franca, A., Nachtergaele, J., & Govers, G. (1999). Concentrated flow erosion rates as affected by rock fragment cover and initial soil moisture content. *Catena*, 36, 315–329.

Poesen, J., Nachtergaele, J., Verstraeten, G., & Valentin, C. (2003). Gully erosion and environmental change: Importance and research needs. *Catena*, 50, 91–133. https://doi.org/10.1016/S0341-8162(02)00143-1

Poesen, J., & Savat, J. (1981). Detachment and transportation of loose sediments by raindrop splash: Part II Detachability and transport ability measurements. *Catena*, 8(1), 19–41.

Poesen, J. W., Torri, D., & Bunte, K. (1994). Effects of rock fragments on soil erosion by water at different spatial scales: A review. *Catena*, 23, 141–166. https://doi.org/10.1016/0341-8162(94)90058-2

Posner, A. J., & Georgakakos, K. P. (2015). Normalized landslide index method for susceptibility map development in El Salvador. Natural Hazards, 79(3), 1825–1845. https://doi.org/10.1007/s11069-015-1930-4

Puustinen, M., Tattari, S., Koskiaho, J., & Linjama, J. (2007). Influence of seasonal and annual hydrological variations on erosion and phosphorus transport from arable areas in Finland. Soil & Tillage Research, 93, 44–55.

Quinton, J. N., Catt, J. A., Wood, G. A., & Steer, J. (2006). Soil carbon losses by water erosion: Experimentation and modeling at field and national scales in the UK. Agriculture, Ecosystems & Environment, 112, 87–102. https://doi.org/10.1016/j.agee.2005.07.005

Rādoane, M., & Rādoane, N. (2005). Dams, sediment sources and reservoir silting in Romania. Geomorphology, 71, 112–125.

Rai, S. C., & Sharma, E. (1998). Comparative assessment of runoff characteristics under different land use patterns within a Himalayan watershed. Hydrological Processes, 12(13), 2235–2248.

Randerson, J. T., Chen, Y., van derWerf, G. R., Rogers, B. M., & Morton, D. C. (2012). Global burned area and biomass burning emissions from small fires. *Journal of Geophysical Research*, *117*, G04012. https://doi.org/10.1029/2012JG002128

Rapp, A., Axelsson, V., Berry, L., & Rust, D. H. M. (1972). Soil erosion and sediment transport in the Morogoro catchment Tanzania. Geografiska Annaler: Series A, Physical Geography, 54, 125–155.

Rapp, I. (1998). Effects of soil properties and experimental conditions on the rill erodibilities of selected soils (PhD thesis). Pretoria, South Africa: Faculty of Biological and Agricultural Sciences, University of Pretoria.

Raymond, P. A., Hartmann, J., Lauerwald, R., Sobek, S., McDonald, C., Hoover, M., . . . Guth, P. (2013). Global carbon dioxide emissions from inland waters. *Nature*, *503*(7476), 355–359. https://doi.org/10.1038/nature12760

Raymond, P. A., Saiers, J. E., & Sobczak, W. V. (2016). Hydrological and biogeochemical controls on watershed dissolved organic matter transport: Pulse-shunt concept. *Ecology*, 97, 5–16. https://doi.org/10.1890/14-1684.1

Regnier, P., Friedlingstein, P., Ciais, P., Mackenzie, F. T., Gruber, N., Janssens, I. A., ... Thullner, M. (2013). Anthropogenic perturbation of the carbon fluxes from land to ocean. *Nature Geoscience*, *6*, 597–607. https://doi.org/10.1038/ngeo1830

Regüés, D., & Gallart, F. (2004). Seasonal patterns of runoff and erosion responses to simulated rainfall in a badland area in Mediterranean mountain conditions (Vallcebre, southeastern Pyrenees). *Earth Surface Processes and Landforms*, *29*, 755–767.

Rejman, J., & Rodzik, J. (2006). Poland. In J. Boardman & J. Poesen (Eds.), Soil erosion in Europe (pp. 95–106). Chichester, UK: John Wiley. Renard, K. G., Foster, G. R., Weesies, G. A., McCool, D. K., & Yoder, D. C. (1997). Predicting soil erosion by water: A guide to conservation planning with the Revised Soil Loss Equation (RUSLE). In Agriculture handbook no. 703 (404 pp.). Washington, DC: U.S. Dept. of Agriculture.

Renschler, C. S., & Harbor, J. (2002). Soil erosion assessment tools from point to regional scales—The role of geomorphologists in land management research and implementation. *Geomorphology*, 47, 189–209. https://doi.org/10.1016/S0169-555X(02)00082-X Restrepo, J. D., Kjerfve, B., Hermelin, M., & Restrepo, J. C. (2006). Factors controlling sediment yield in a major South American drainage

basin: The Magdalena River, Colombia. Journal of Hydrology, 316, 213–232.

Richey, J. E., Krusche, A. V., Johnson, M. S., Cunha, H. B., & Ballester, M. V. (2009). The role of rivers in the regional carbon balance. *Geophysical Monograph Series*, 186, 489–504.

Rodríguez-Blanco, M. L., Taboada-Castro, M. M., & Taboada-Castro, M. T. (2013). Linking the field to the stream: Soil erosion and sediment yield in a rural catchment, NW Spain. Catena, 102, 74–81.

Roering, J. J., Kirchner, J. W., & Dietrich, W. E. (1999). Evidence for nonlinear, diffusive sediment transport on hillslopes and implications for landscape morphology. *Water Resources Research*, 35, 853–870. https://doi.org/10.1029/1998WR900090

Sadeghi, S. H. R., Seghaleh, M. B., & Rangavar, A. S. (2013). Plot sizes dependency of runoff and sediment yield estimates from a small watershed. Catena, 102, 55–61.

Schröder, W., & Theune, C. (1984). Feststoffabtrag und Stauraumverlandung in Mitteleuropa. Wasserwirtschaft, 74, 374–379.

Scott, D. F., Versfeld, D. B., & Lesch, W. (1998). Erosion and sediment yield in relation to afforestation and fire in the mountains of the Western Cape Province, South Africa. South African Geographical Journal, 80, 52–59.

Sharma, P. S. C. R., & Rai, S. C. (2004). Streamflow, sediment and carbon transport from a Himalayan watershed. Journal of Hydrology, 289(1), 190–203.

Sheffield, J., Goteti, G., & Wood, E. F. (2006). Development of a 50-yr high-resolution global dataset of meteorological forcings for land surface modeling. Journal of Climate, 19(13), 3088–3111.

Sheffield, J., & Wood, E. F. (2007). Characteristics of global and regional drought, 1950–2000: Analysis of soil moisture data from off-line simulation of the terrestrial hydrologic cycle. Journal of Geophysical Research, 112, D17115. https://doi.org/10.1029/2006JD008288

Shibata, H., Mitsuhashi, H., Miyake, Y., & Nakano, S. (2001). Dissolved and particulate carbon dynamics in a cool-temperate forested basin in northern Japan. *Hydrological Processes*, 15(10), 1817–1828.

Sickman, J. O., Zanoli, M. J., & Mann, H. L. (2007). Effects of urbanization on organic carbon loads in the Sacramento River, California. Water Resources Research, 43, W11422. https://doi.org/10.1029/2007WR005954

Sine, L., & Agneessens, J. (1978). Etude des débits solides et du phénomène de migration dans une rivière drainant un bassin agricole. Pédologie, 28, 183–191.

Sisák, I., Máté, F., & Szücs, P. (2007). Sediment and phosphorous loads from three small catchments in the watershed of Lake Balaton. Paper presented at the proceedings of the COST 634 international conference on off-site impacts of soil erosion and sediment transport. Prague, Czech Republic.

Smith, J. C., Galy, A., Hovius, N., Tye, A. M., Turowski, J. M., & Schleppi, P. (2013). Runoff-driven export of particulate organic carbon from soil in temperate forested uplands. *Earth and Planetary Science Letters*, 365, 198–208. https://doi.org/10.1016/j.epsl.2013.01.027

Snedecor, G. W., & Cochran, W. (1989). *Statistical methods* (8th ed.). Iowa City: Iowa State University Press.

Sogon, S., Penven, M., Bonte, P., & Muxart, T. (1999). Estimation of sediment yield and soil loss using suspended sediment load and ¹³⁷Cs measurements on agricultural land, Brie Plateau, France. *Hydrobiologia*, 410, 251–261.

Soler, M., Latron, J., & Gallart, F. (2008). Relationships between suspended sediment concentrations and discharge in two small research basins in a mountainous Mediterranean area, Vallcebre (Eastern Pyrenees). *Geomorphology*, 98, 143–152.

Steegen, A. (2001). Sediment deposition in and export from small agricultural catchments (PhD thesis, 220 pp.). Leuven, Belgium: Faculty of Sciences, Geography, Katholieke Universiteit Leuven.

Stone, M., & Saunderson, H. C. (1996). Regional patterns of sediment yield in the Laurentian Great Lakes basin. IAHS Publication, 126, 125– 131.

Stutter, M. I., Deeks, L. K., Low, D., & Billett, M. F. (2006). Impact of soil and groundwater heterogeneity on surface water chemistry in an upland catchment. *Journal of Hydrology*, 318, 103–120.

Syvitski, J. P. M., & Milliman, J. D. (2007). Geology, geography, and humans battle for dominance over the delivery of fluvial sediment to the coastal ocean. *The Journal of Geology*, *115*, 1–19. https://doi.org/10.1086/509246

Syvitski, J. P. M., Vörösmarty, C. J., Kettner, A. J., & Green, P. (2005). Impact of humans on the flux of terrestrial sediment to the global coastal ocean. Science, 308, 376–380. https://doi.org/10.1126/science.1109454

Tamburino, V., Barbagallo, S., & Vella, P. (1990). Evaluation of sediment deposition in Sicilian artificial reservoirs. *IAHS Publication*, 194, 107–112.

Temme, A. J. A. M., & Vanwalleghem, T. (2016). LORICA—A new model for linking landscape and soil profile evolution: Development and sensitivity analysis. *Computers & Geosciences*, 90, 131–143. https://doi.org/10.1016/j.cageo.2015.08.004

Tesfa, T. K., Tarboton, D. G., Watson, D. W., Schreuders, K. A. T., Baker, M. E., & Wallace, R. M. (2011). Extraction of hydrological proximity measures from DEMs using parallel processing. *Environmental Modelling & Software*, 26(12), 1696–1709.

Thodsen, H., Hasholt, B., & Perjup, M. (2004). Transport of phosphorus, wash load and suspended sediment in the River Varde Å in southwest Jutland, Denmark. IAHS Publication, 288, 466–473.

Torri, D., Borselli, L., Guzzetti, F., Calzolari, M. C., Bazzofi, P., Ungaro, F., . . . Salvador Sanchis, M. P. (2006). Italy. In J. Boardman & J. Poesen (Eds.), Soil erosion in Europe (pp. 245–262). Chichester, UK: John Wiley.

Tropeano, D. (1991). High flow events and sediment transport in small streams in the 'tertiary basin' area in piedmont (Northwest Italy). *Earth Surface Processes and Landforms*, 16, 323–339. https://doi.org/10.1002/esp.3290160406

Tucker, G. E., & Bras, R. L. (2000). A stochastic approach to modeling the role of rainfall variability in drainage basin evolution. Water Resources Research, 36, 1953–1964.

Valero-Garcés, B. L., Navas, A., Machín, J., & Walling, D. (1999). Sediment sources and siltation in mountain reservoirs: A case study from the Central Spanish Pyrenees. *Geomorphology*, 28, 23–41.

van Dijk, A. I. J. M., & Bruijnzeel, L. A. (2001). Modelling rainfall interception by vegetation of variable density using an adapted analytical model. Part 1. Model description. *Journal of Hydrology*, 247, 230–238.

Van Dijk, P. M., & Kwaad, F. J. P. M. (1996). Runoff generation and soil erosion in small agricultural catchments with loess-derived soils. *Hydrological Processes*, 10, 1049–1059.

Van Hoestenberghe, T., Voet, Eylenbosch, M. J., & Cabus, P. (2008). Sedimentexport door onbevaarbare waterlopen in Vlaanderen (D/2008/ 6871/021, 131 pp.). Metingen 2003–2007, Vlaamse Milieumaatschappij, Afdeling Operationeel Waterbeheer, Belgium.

Vanmaercke, M., Kettner, A. J., Van Den Eeckhaut, M., Poesen, J., Mamaliga, A., Verstraeten, G., ... Govers, G. (2014a). Moderate seismic activity affects contemporary sediment yields. *Progress in Physical Geography*, 38(2), 145–172. https://doi.org/10.1177/ 0309133313516160

Vanmaercke, M., Poesen, J., Broeckx, J., & Nyssen, J. (2014b). Sediment yield in Africa. Earth-Science Reviews, 136, 350–368. https://doi.org/ 10.1016/j.earscirev.2014.06.004

Vanmaercke, M., Poesen, J., Govers, G., & Verstraeten, G. (2015). Quantifying human impacts on catchment sediment yield: A continental approach. *Global and Planetary Change*, 130, 22–36. https://doi.org/10.1016/j.gloplacha.2015.04.001

Vanmaercke, M., Poesen, J., Van Mele, B., Demuzere, M., Bruynseels, A., Golosov, V. . . . Yermolaev, O. (2016). How fast do gully headcuts retreat? *Earth-Science Reviews*, 154, 336–355.

Vanmaercke, M., Poesen, J., Verstraeten, G., de Vente, J., & Ocakoglu, F. (2011). Sediment yield in Europe: Spatial patterns and scale dependency. Geomorphology, 130, 142–161.

Vannote, R. L., Minshall, G. W., Cummins, K. W., Sedell, J. R., & Cushing, C. E. (1980). The river continuum concept. Canadian Journal of Fisheries and Aquatic Sciences, 37, 130–137. https://doi.org/10.1139/f80-017

Van Rompaey, A., Bazzoffi, P., Jones, R. J. A., & Montanarella, L. (2005). Modelling sediment yields in Italian catchments. Geomorphology, 65, 157–169.

van Wesemael, B., Poesen, J., & de Figueiredo, T. (1995). Effects of rock fragments on physical degradation of cultivated soils by rainfall. Soil & Tillage Research, 33, 229–250. https://doi.org/10.1016/0167-1987(94)00439-L

Verachtert, E., Maetens, W., Van Den Eeckhaut, M., Poesen, J., & Deckers, J. (2011). Soil loss rates due to piping erosion. Earth Surface Processes and Landforms, 36, 1715–1725.

Verstraeten, G. (2000). Modderoverlast, sedimentatie in wachtbekkens en begroting van de sedimentexport naar waterlopen in Midden-België (PhD thesis). Leuven, Belgium: Katholieke Universiteit Leuven.

Verstraeten, G., & Poesen, J. (2001). Factors controlling sediment yield from small intensively cultivated catchments in a temperate humid climate. *Geomorphology*, 40, 123–144.

Verstraeten, G., & Poesen, J. (2002). Regional scale variability in sediment and nutrient delivery from small agricultural watersheds. Journal of Environmental Quality, 31, 870–879.

Verstraten, J. M. (1977). Chemical erosion in a forested watershed in the oesling, Luxembourg. Earth-Surface Processes, 2, 175–184. https://doi.org/10.1002/esp.3290020209

Vihermaa, L. E., Waldron, S., Domingues, T., Grace, J., Cosio, E. G., Limonchi, F., ... Gloor, E. (2016). Fluvial carbon export from a lowland Amazonian rainforest in relation to atmospheric fluxes. *Journal of Geophysical Research: Biogeosciences*, 121, 3001–3018. https://doi.org/ 10.1002/2016JG003464

Vihermaa, L. E., Waldron, S., Garnett, M. H., & Newton, J. (2014). Old carbon contributes to aquatic emissions of carbon dioxide in the Amazon. *Biogeosciences*, 11, 3635–3645. https://doi.org/10.5194/bg-11-3635-2014

Voet, M. (1997). Sedimentafvoeren in kleine stroomgebieden. Water, 92, 10-14.

Vuorenmaa, J., Rekolainen, S., Lepisto, A., Kenttamies, K., & Kauppila, P. (2002). Losses of nitrogen and phosphorus from agricultural and forest areas in Finland during the 1989s and 1990s. Environmental Monitoring and Assessment, 76, 213–248.

Wallis, C., Watson, D. W., Tarboton, D. G., & Wallace, R. (2009). *Hydrologic terrain processing using parallel computing*. Paper presented at the 18th world IMACS/MODSIM congress. Cairns, Australia.

Ward, N. D., Krusche, A. V., Sawakuchi, H. O., Brito, D. C., Cunha, A. C., Moura, J. M. S., . . . Richey, J. E. (2015). The compositional evolution of dissolved and particulate organic matter along the lower Amazon River-Óbidos to the ocean. *Marine Chemistry*, 177, 244–256.

Warrick, J., Melack, J. M., & Goodridge, B. M. (2015). Sediment yields from small, steep coastal watersheds of California. *Journal of Hydrology*, 4, 516–534.

Warrick, J. A., Hatten, J. A., Pasternack, G. B., Gray, A. B., Goni, M. A., & Wheatcroft, R. A. (2012). The effects of wildfire on the sediment yield of a coastal California watershed. *GSA Bulletin*, 124(7–8), 1130–1146. https://doi.org/10.1130/B30451.1

Watson, R. T., Noble, I. R., Bolin, B., Ravindranath, N. H., Verardo, D. J., & Dokken, D. J. (2000). Land use, land-use change, and forestry. In Intergovernmental Panel on Climate Change special report (375 pp.). Cambridge, UK: Cambridge University Press.

West, T. O., Brandt, C. C., Baskaran, L. M., Hellwinckel, C. M., Mueller, R., Bernacchi, C. J., ... Post, W. M. (2010). Cropland carbon fluxes in the United States: Increasing geospatial resolution of inventory-based carbon accounting. *Ecological Applications*, 20, 1074–1086. https:// doi.org/10.1890/08-2352.1

White, S. (1996). Erosion, sediment delivery and sediment yield patterns in the Philippines. IAHS Publication, 236, 233-240.

Yang, J., He, Y., Aubrey, D. P., Zhuang, Q., & Teskey, R. O. (2016). Global patterns and predictors of stem CO₂ efflux in forest ecosystems. *Global Change Biology*, 22, 1433–1444. https://doi.org/10.1111/gcb.13188

Yu, B., Sajjapongse, A., Yin, D., Eusof, Z., Anecksamphant, C., Rose, C. W., & Cakurs, U. (1999). Application of a physically based soil erosion model, GUEST, in the absence of data on runoff rates-II. Four case studies from China, Malaysia and Thailand. Australian Journal of Soil Research, 37, 13–31.